%Preamble

\documentclass{amsart}
\usepackage{amssymb}
\usepackage[mathscr]{eucal}
\usepackage{eufrak}
\usepackage{xspace}
\usepackage[latin1]{inputenc}

\textwidth=6.5in
\oddsidemargin=0in
\evensidemargin=0in
\topmargin=0in

%user defined commands

\newcommand{\HH}{\mathcal H}
\renewcommand{\SS}{\mathbb{S}}
\newcommand{\yy}{\mathbf{y}}
\newcommand{\ff}{\mathbf{f}}
\newcommand{\qq}{\mathbf{q}}
\newcommand{\bb}{\mathbf{b}}
\newcommand{\hh}{\mathbf{h}}
\newcommand{\xix}{\boldsymbol{\xi}}
\newcommand{\zz}{\mathbf{z}}
\newcommand{\gzz}{\boldsymbol{\zeta}}
\newcommand{\xci}{\overline{\xx}{}^i}
\newcommand{\xfi}{\,{}^i\overline{\xx}}
\newcommand{\xici}{\overline{\xix}{}^i}
\newcommand{\xifi}{\,{}^i\overline{\xix}}
\newcommand{\xc}[2]{\overline{#1}{}^{#2}}
\newcommand{\xf}[2]{\,{}^{#2}\overline{#1}}
\newcommand{\gO}{\Omega}
\newcommand{\gz}{\zeta}
\newcommand{\R}{\mathbb{R}}
\newcommand{\Q}{\mathbb{Q}}
\newcommand{\D}{\mathbb{D}}
\newcommand{\Z}{\mathbb{Z}}
\newcommand{\N}{\mathbb{N}}
\newcommand{\C}{\mathbb{C}}
\newcommand{\LL}{\mathcal{L}}
\newcommand{\NN}{\mathcal{N}}
\newcommand{\FF}{\mathcal{F}}
\newcommand{\GG}{\mathcal{G}}
\newcommand{\GN}{\mathcal{G}^N}
\newcommand{\GB}{{\mathbb G}}
\newcommand{\RR}{\mathcal{R}}
\newcommand{\QQ}{\mathcal{Q}}
\newcommand{\PP}{\mathcal{P}}
\newcommand{\KK}{\mathbf{K}}
\renewcommand{\H}{\mathcal{H}}
\newcommand{\CS}{\mathcal{S}}
\newcommand{\CT}{\mathcal{T}}
\newcommand{\CI}{\mathcal{I}}
\newcommand{\E}{{\mathcal E}}
\newcommand{\U}{{\mathcal U}}
\newcommand{\OO}{{\mathcal O}}
\newcommand{\xx}{\mathbf{x}}
\newcommand{\ga}{\alpha}
\newcommand{\gb}{\beta}
\newcommand{\gc}{\gamma}
\newcommand{\gD}{\Delta}
\newcommand{\go}{\omega}
\newcommand{\gs}{\sigma}
\newcommand{\gS}{\Sigma}
\newcommand{\gL}{\Lambda}
\newcommand{\po}{\partial}
\newcommand{\Dom}{[0,\ell)\times[0,\infty)}
\newcommand{\nuxt}{\nu_{\!x,t}}
\newcommand{\wto}{\rightharpoonup}
\newcommand{\ve}{\varepsilon}
\newcommand{\muTx}{\mu_T^\xi}
\newcommand{\mux}{\mu^\xi}
\newcommand{\te}{\theta}
\newcommand{\T}{\Theta}
\newcommand{\la}{\langle}
\newcommand{\ra}{\rangle}
\newcommand{\gd}{\delta}
\newcommand{\nux}{\nu^\xi}
\newcommand{\gl}{\lambda}
\newcommand{\loc}{{\text{loc}}}
\newcommand{\X}{\times}
\newcommand{\del}{{\partial}}
\renewcommand{\O}{{\mathbf O}}
\renewcommand{\d}{\delta}
\renewcommand{\l}{\lambda}
\newcommand{\lb}{\lambda}
\renewcommand{\a}{\alpha}
\renewcommand{\b}{\beta}
\renewcommand{\t}{\theta}
\newcommand{\s}{\sigma}
\newcommand{\g}{\gamma}
\renewcommand{\k}{\kappa}
\newcommand{\Tr}{{\!\top}}
\newcommand{\Om}{\Omega}
\newcommand{\om}{\omega}
\newcommand{\mT}{\mu_T^\xi}
\newcommand{\mx}{\mu^\xi}
\newcommand{\supp}{\text{\rm supp}\,}
\newcommand{\re}{\mathbb{R}}
\newcommand{\M}{{\mathcal M}}
\newcommand{\V}{{\mathcal V}}
\newcommand{\DM}{\mathcal D\mathcal M}
\renewcommand{\ni}{\noindent}
\renewcommand{\div}{\text{\rm div}\,}
\newcommand{\grad}{\nabla}
\newcommand{\divx}{\text{\rm div}_x}
\newcommand{\gradx}{\nabla_x}
\newcommand{\wlim}{\text{w-}\lim}
\newcommand{\essl}{\text{ess}\lim}
\renewcommand{\P}[2]{\Phi\bigl(#1\mid #2\bigr)}
\newcommand{\bW}{\overline{W}}
\newcommand{\bS}{\overline{S}}
\newcommand{\bu}{\overline{u}}
\newcommand{\bv}{\overline{v}}
\newcommand{\bp}{\overline{p}}
\newcommand{\bOm}{\overline{\Om}}
\newcommand{\tr}{\tilde r}
\newcommand{\tl}{\tilde l}
\newcommand{\tW}{\widetilde{W}}
\newcommand{\tS}{\widetilde{S}}
\newcommand{\tu}{\tilde{u}}
\newcommand{\tv}{\tilde{v}}
\newcommand{\tp}{\tilde{p}}
\newcommand{\tU}{\widetilde{U}}
\newcommand{\ubf}{\mathbf{u}}
\newcommand{\vbf}{\mathbf{v}}
\newcommand{\zbf}{\mathbf{z}}
\newcommand{\wbf}{\mathbf{w}}
\newcommand{\fbf}{\mathbf{f}}
\newcommand{\gbf}{\mathbf{g}}
\newcommand{\hbf}{\mathbf{h}}

\newcommand{\vph}{\varphi}

\newcommand{\Fbf}{\mathbf{F}}
\newcommand{\Gbf}{\mathbf{G}}
\newcommand{\Abf}{\mathbf{A}}
\newcommand{\Bbf}{\mathbf{B}}
\newcommand{\Ibf}{\mathbf{I}}
\newcommand{\Tbf}{\mathbf{T}}
\newcommand{\Lip}{\text{\rm Lip}}
\newcommand{\Sbb}{\mathbb S}
\renewcommand{\S}{{\mathcal S}}
\renewcommand{\subset}{\subseteq}
\newcommand{\e}{\epsilon}
\newcommand{\bubf}{\overline{\ubf}}
\newcommand{\Mx}{M_x}
\newcommand{\Sap}{$S$-$a.p.$\ }
\newcommand{\K}{\mathcal{K}}
\newcommand{\Nu}{\mathcal{V}}
\newcommand{\TV}{\text{TV}\,}
\newcommand{\BV}{\operatorname{BV}}
\newcommand{\MM}{\mathbf{M}}
\newcommand{\BUC}{\operatorname{BUC}}
\newcommand{\BC}{\operatorname{BC}}
\newcommand{\AP}{\operatorname{AP}}
\newcommand{\BAP}{\operatorname{BAP}}
\newcommand{\util}[1]{\underline{#1}}
\newcommand{\DD}{{\mathcal D}}
\newcommand{\BB}{{\mathcal B}}
\renewcommand{\AA}{{\mathcal A}}
\newcommand{\BBAP}{\BB{\AP}}
\newcommand{\BBAp}{\BB'{\AP}}
\newcommand{\PC}{\operatorname{PC}}
\newcommand{\tg}{\tilde{g}}
\newcommand{\XX}{{\mbox{\boldmath$X$}}}
\newcommand{\intm}[1]{{\int_{#1}\!\!\!\!\!\!\!\!\!\!\!-}\,}
\newcommand{\medint}{{\mbox{\vrule height3.5pt depth-2.8pt
          width4pt}\mkern-13mu\int\nolimits}}
\newcommand{\Medint}{\mkern12mu\mbox{\vrule height4pt
         depth-3.2pt
          width5pt}\mkern-16.5mu\int\nolimits}
\newcommand{\eqa}[2]{\left(\eta_{#1}^{++}(u,u_{#2}),\qq_{#1}^{++}(u,u_{#2})\right)}
\newcommand{\eqb}[2]{\left(\eta_{#1}^{0+}(u,u_{#2}),\qq_{#1}^{0+}(u,u_{#2})
\right)}
\newcommand{\eqc}[2]{\left(\eta_{#1}^{+0}(u,u_{#2}),\qq_{#1}^{+0}(u,u_{#2})
\right)}
\newcommand{\sgn}{\operatorname{sgn}}
\renewcommand{\supp}{\operatorname{supp}}
\newcommand{\jz}{{j_0}}

\theoremstyle{plain}
\newtheorem{theorem}{Theorem}[section]
\newtheorem{corollary}{Corollary}[section]
\newtheorem*{main}{Main Theorem}
\newtheorem{lemma}{Lemma}[section]
\newtheorem{proposition}{Proposition}[section]
\theoremstyle{definition}
\newtheorem{definition}{Definition}[section]
\theoremstyle{remark}
\newtheorem*{notation}{Notation}
\newtheorem{remark}{Remark}[section]
\numberwithin{equation}{section}
%% Typset with custart format


\begin{document}
\begin{thebibliography}{99}

\bibitem{BOP1} D.~Bekiranov, T.~Ogawa and G.~Ponce. {\sl Weak solvability and well-posedness of a coupled Schrödinger-Korteweg De Vries equation for capillary-gravity wave interactions}. Proceedings of the American Mathematical Society {\bf125}, No.~10 (1997), 2907--2919.

\bibitem{BOP2} D.~Bekiranov, T.~Ogawa and G.~Ponce. {\sl Interaction equations for short and long dispersive waves}. 
Journal of Functional Analysis {\bf 158} (1998), 357--388. 

\bibitem{Be} D.J.~Benney. {\sl A general theory for interactions between short and long waves}. Studies in Applied Mathematics {\bf 56} (1977), 81--94. 

\bibitem{Br} A.~Bressan. ``Hyperbolic Systems of Conservation Laws. The one-dimensional Cauchy problem''. Oxford Lecture Series in Mathematics and its Applications, {\bf20}. Oxford University Press, Oxford, 2000. xii+250 pp. 

\bibitem{CK1} G.-Q.~Chen and P.-T.~Kan. {\sl Hyperbolic conservation laws with umbilic degeneracy. I}. Arch.\ Rational Mech.\ Anal.\ {\bf130} (1995), no. 3, 231--276.


\bibitem{CK2} G.-Q.~Chen and P.-T.~Kan. {\sl Hyperbolic conservation laws with umbilic degeneracy}. Arch.\ Ration.\ Mech.\ Anal.\ {\bf 160} (2001), no. 4, 325--354.

\bibitem{Da} C.M.~Dafermos. ``Hyperbolic Conservation Laws in Continuum Physics''. Grundlehren der Mathematischen Wissenschaften [Fundamental Principles of Mathematical Sciences], 325. Springer-Verlag, Berlin, 2000. xvi+443 pp. 

\bibitem{DFig} J.P.~Dias and M.~Figueira. {\sl Existence of weak solutions for a quasilinear version of Benney equations}. Journal of Hyperbolic Differential Equations {\bf 4} (2007), 555-563.

\bibitem{DFigO} J.P.~Dias, M.~Figueira and F.~Oliveira. {\sl Existence of local strong solutions for a quasilinear Benney system}. C.~R.~Acad.~Sci.~Paris, Ser.~1 {\bf344} (2007), 493--496.



\bibitem{DF} J.P.~Dias and H.~Frid. {\sl Radially symmetric weak solutions for a quasilinear wave equation in two space dimensions}. J.~Differential Equations {\bf 219} (2005), no. 2, 306--322.

\bibitem{FS1} H.~Frid and M.~Santos. {\sl Non-strictly hyperbolic systems of conservation laws of the conjugate type}. Comm.\ Partial Differential Equations {\bf19} (1994), no. 1-2, 27--59. 

\bibitem{FS2} H.~Frid and M.~Santos. {\sl The Cauchy problem for the systems $\partial\sb t z-\partial\sb x(\overline z{}\sp \gamma)=0$}. J.~Differential Equations {\bf111} (1994), no. 2, 340--359.

\bibitem{HR} H.~Holden and N.H.~Risebro. ``Front Tracking for Hyperbolic Conservation Laws''. Applied Mathematical Sciences, 152. Springer-Verlag, New York, 2002. xii+363 pp.

\bibitem{KK} B.~Keyfitz and H.~Kranzer. {\sl The Riemann problem for a class of hyperbolic conservation laws exhibiting a parabolic degeneracy}. J.~ Differential Equations {\bf47} (1983), no. 1, 35--65.

\bibitem{LF} Ph.~LeFloch. ``Hyperbolic Systems of Conservation Laws. The theory of classical and nonclassical shock waves''. Lectures in Mathematics ETH Zürich. Birkhäuser Verlag, Basel, 2002. x+294 pp.

\bibitem{L} J.-L.~Lions. ``Quelques Méthodes de Résolution des Problèmes aux Limites Non Linéaires''. (French) Dunod; Gauthier-Villars, Paris 1969 xx+554 pp. 

\bibitem{LM} J.-L.~Lions and E.~Magenes. ``Probl\`emes aux Limites Non Homog\`enes et Applications''. Vol.~1. Dunod, Paris, 1968. 
  

\bibitem{Mu} F.~Murat. {\sl Compacité par compensation}. (French) Ann.\ Scuola Norm.\ Sup.\ Pisa Cl.\ Sci.\ (4) {\bf5} (1978), no.~3, 489--507. 

\bibitem{Ra} J.~Rauch. ``Partial Differential Equations''. Springer, 1992.

\bibitem{Se1} D.~Serre. ``Systems of Conservation Laws. 2. Geometric structures, oscillations, and initial-boundary value problems''. Translated from the 1996 French original by I. N. Sneddon. Cambridge University Press, Cambridge, 2000. xii+269 pp.

\bibitem{Se2} D.~Serre. {\sl Les ondes planes en electromagnetisme non lineaire}. Physica D {\bf31} (1988), 227--251.  

\bibitem{SeSh} D.~Serre and J.~Shearer. {\sl Convergence with physical viscosity for nonlinear elasticity}. Unpublished
preprint (1993), available from D. Serre under request.

\bibitem{T} Tartar, L. {\sl Compensated compactness and applications to partial differential equations}. Nonlinear analysis and mechanics: Heriot-Watt Symposium, Vol. IV, pp. 136--212, Res.\ Notes in Math., {\bf39}, Pitman, Boston, Mass.-London, 1979.

\bibitem{TH1} M.~Tsutsumi and S.~Hatano. {\sl Well-posedness of the Cauchy problem for the long wave-short wave resonance equations}. Nonlinear Analysis, Theory, Methods \& Applications, {\bf 22} (1994), No.~2, 155--171.


\bibitem{TH2} M.~Tsutsumi and S.~Hatano. {\sl Well-posedness of the Cauchy problem for Benney's first equations of long wave short wave interactions}. Funkcialaj Ekvacioj, {\bf37} (1994), 289--316. 
  

\end{thebibliography}


\title[Short vs.\ long wave interactions for systems of conservation laws]
{Vanishing viscosity with short wave long wave interactions for systems of conservation laws}
\author{Jo\~ao Paulo Dias}
\address{CMAF/UL and FC/UL, Universidade de Lisboa, Av.\ Gama Pinto, 2, 1649-003 Lisboa, Portugal}
\email{dias@ptmat.fc.ul.pt}
\author{M\'ario Figueira}
\address{CMAF/UL and FC/UL, Universidade de Lisboa, Av.\ Gama Pinto, 2, 1649-003 Lisboa, Portugal}
\email{figueira@ptmat.fc.ul.pt}
\author{Hermano Frid}
\address{Instituto de Matem\'atica Pura e Aplicada - IMPA Estrada Dona Castorina, 110
22460-320, Rio de Janeiro, RJ,  Brazil}
\email{hermano@impa.br}
\keywords{short wave long wave interaction, systems of conservation laws, vanishing viscosity}
\subjclass{35L65, 35L70}
\date{}

\begin{abstract} Motivated by Benney's general theory, we propose new models for short wave long wave interactions when the long waves are described by nonlinear systems of conservation laws. We prove  the strong convergence of the solutions of the vanishing viscosity and short wave long wave interactions systems by using compactness results from the compensated compactness theory and new  energy estimates obtained for the coupled systems. We analyse several of the representative examples such as scalar conservation laws, general symmetric systems, nonlinear elasticity, nonlinear electromagnetism.


\end{abstract}

\maketitle

\section{Introduction} \label{S:0}

Motivated by Benney \cite{Be}, we introduce here several examples of systems of conservation laws  coupled with 
a semilinear Schr\"odinger equation modelling  short wave long wave (SW-LW) interactions for a broad class of physical problems. Several systems motivated by Benney's general theory for SW-LW interactions have appeared in the literature.
The case where long waves are described by linear equations was considered in, e.g., \cite{TH1,TH2}. The case where long waves are
described by the KdV  and other dispersive equations was considered in, e.g., \cite{BOP1,BOP2}. 

Here we formulate a general framework for SW-LW interactions when long waves are described by a nonlinear system of conservation laws. In order to upgrade Benney's theory to allow interactions involving nonlinear systems we generalize 
the scope of that theory and for each  particular class of systems we propose the corresponding nonlinear SW-LW interaction coupling based on its own characteristic structure. 

We now briefly describe the models of SW-LW interactions considered in this paper, which cover  most of the well known examples of systems of conservation laws. For all these models we study the convergence of the vanishing viscosity method. In general, the models present a viscosity parameter $\ve$ and an {\em interaction parameter} $\a$.
In the first three applications we will be concerned with the convergence of the  sequence of solutions when the viscosity $\ve$ goes to 0, while the interaction parameter $\a$ is kept fixed.  The limit of the vanishing viscosity solutions is a weak solution of the Cauchy problem for a system
of Benney's type presenting a nonlinear coupling between the Schrödinger equation and the respective nonlinear hyperbolic system of conservation laws.

In the last three examples we analise the convergence of the solutions when both viscosity and interaction parameters, $\ve,\a$, go to zero with $\a=O(\ve^{1/2})$. In particular, for these  examples, in the limit $\ve=0$ the Schrödinger equation decouples from the respective nonlinear system of conservation laws and the limit of the vanishing viscosity solutions is a weak solution of the decoupled system. This decoupling is in fact a more realistic situation which is in accordance with the fact that shock wave is a macroscopic phenomenon while Schrödinger equation describes a microcopic
one.  



\subsection{Scalar conservation laws} To warm up, we begin with the simplest case, that is, scalar conservation laws. 
In this case our general approach proposes the following system:
\begin{align}
& iu_t+u_{xx}=u|u|^2+\a g(v)u,\label{e1.1}\\
&v_t+f(v)_x=\a (g'(v)|u|^2)_x +\ve v_{xx}\label{e1.2}.
\end{align}
The new ingredient here, comparing to the original Benney's formulation, is the appearance of a general function $g(v)$ ({\em interaction function}) in the Schrödinger equation and the corresponding derivative $g'(v)$ in the scalar conservation law, instead of $v$ and $1$ prescribed by that formulation. If one chooses $g(v)$ such that $g'(v)$ has a suitable support, the new system preserve physically significant domains. For instance, if $v$ describes a density or a concentration, setting $g(v)=v$ may lead to solutions with no physical meaning taking negative values. In such cases, it is natural to replace $v$ by a suitable function, say, a smooth approximation of $(v)_+$, such that $\supp g'\subset[0,\infty)$, which we easily prove to be sufficient to guarantee the nonnegativity of the solution. Thus, choosing $g(v)$ that preserves physically relevant domains we are also able to easily prove the strong convergence of the vanishing viscosity solutions as $\ve\to0$, for any fixed $\alpha>0$, independent of $\ve$. 

We remark that the convergence of the vanishing viscosity solutions of the Cauchy problem for \eqref{e1.1},\eqref{e1.2} for $g(v)=v$, 
in the particular case $f(v)=a\,v^2-b\,v^3$, with $b>0$,  was proved in \cite{DFig}, while
the existence of local in time smooth solution of the Cauchy problem for \eqref{e1.1},\eqref{e1.2},
with $\ve=0$ (hyperbolic case), $g(v)=v$,  and a general flux function $f$ was obtained in \cite{DFigO}. In both of these papers $\a$ is taken $=1$ for simplicity.

\subsection{A degenerate symmetric system}\label{sS:1} As a bridge from the scalar case and more difficult nonlinear systems of conservation laws, we consider the  degenerate symmetric system introduced by Keyfitz and Kranzer \cite{KK}. As is well known, this system, in a vector dependent variable $v$, is basically the combination of a scalar conservation law for
$r=|v|$ with transport equations for the components of $v/r$. For this system we propose the following model of SW-LW interactions 
\begin{align}
& iu_t+u_{xx}=u|u|^2+\a g(r)u,\label{e1.3}\\
& v_t+(\vph(r)v)_x=\a (|u|^2g'(r)\frac{v}r)_x +\ve(r_{x}\frac{v}r)_x,\qquad r=|v|. \label{e1.4} 
\end{align}
Here also, since the interaction function depends only on $r$, our approach allows the invariance of the physically meaningful region $r\ge0$. Based on the just mentioned result for scalar conservation laws we easily obtain the convergence of the vanishing viscosity solutions as $\ve\to0$ for any fixed $\a>0$ independent of $\ve$. 

\subsection{The $p$-system of the nonlinear elasticity (I)}\label{sS:4}  
As an  example involving an authentic  nonlinear system of conservation laws, we consider one of the the most representative of all $2\X2$ systems, namely, that modeling nonlinear elasticity. For this system we propose the following     
for SW-LW interactions
\begin{align} 
& iu_t+u_{xx}=u|u|^2+ \a g(v)u,\label{e1.10}\\
& w_t-\s(v)_x=\a(g'(v)|u|^2)_x +\ve w_{xx},\label{e1.11}\\
&v_t-w_x=0.\label{e1.12}
\end{align}
As is well known, in this case no positively invariant region in the plane is available for the corresponding (Navier-Stokes with nonsingular pressure) system \eqref{e1.11}-\eqref{e1.12}, when the SW-LW interaction term $\a(g'(v)|u|^2)_x$ is omited. In this way, there is a larger flexibility in the choice of the function $g(v)$. In particular, we may choose $g(v)=v$. The lack of bounded invariant domains was surmounted by Serre and Shearer in \cite{SeSh} through an analysis based on $L^p$ estimates with $p<\infty$ and compensated compactness (see also \cite{DF}). In this paper, we show how to get the necessary $L^p$ bounds so as to allow the application of the compactness result in \cite{SeSh}. In this way, for the special choice $g(v)=v$,  we obtain the strong convergence of the solutions with vanishing viscosity   as $\ve\to0$, for any fixed  $\a>0$ independent of $\ve$. 

\subsection{The $p$-system of the nonlinear elasticity (II)}\label{sS:4'} 

We also consider the case where $g(v)$ is a general nonlinear interaction function subjected to suitable nonlinearity restrictions. For such general interaction functions we address the question of the convergence of the solutions of the Cauchy problem for \eqref{e1.10},\eqref{e1.11},\eqref{e1.12} when $\ve$ and $\a$ tend to 0. Under suitable nonlinearity assumptions for $\s$ and $g$, we prove the strong convergence of these solutions when $\ve\to0$ and $\a=O(\ve^{1/2})$.  

\subsection{Nonlinear electromagnetism}\label{sS:2} As a second example of vanishing viscosity and interaction parameters involving an authentic nonlinear $2\X2$ system, we consider the equations from nonlinear eletromagnetism first thoroughly mathematically analyzed in \cite{Se2}. In this case, the original  $6\X6$ system reduces to a $4\X4$ system in the vector dependent variable $v=(v_1,v_2,v_3,v_4)$, combining a nonlinear $2\X2$ system of conservation laws for $p=\sqrt{v_1^2+v_2^2}$ and $q=\sqrt{v_3^2+v_4^2}$ and transport equations for $a=v_1/p$ and $b=v_3/q$. We propose the following model for SW-LW interactions in this case
\begin{align}
& iu_t+u_{xx}=u|u|^2+\a g(r)u, \qquad\qquad\qquad\qquad\qquad\qquad r=|v|=\sqrt{v_1^2+v_2^2+v_3^2+v_4^2}\label{e1.5}\\
& p_t+(\vph(r)p)_x=-\a(g'(r)\frac{p}r|u|^2)_x+\ve p_{xx},\qquad a_t+(\vph(r)+\a\frac{g'(r)}r|u|^2-\ve p_x)a_x=0, \quad p=\sqrt{v_1^2+v_2^2}\label{e1.6}\\
& q_t-(\vph(r)q)_x=\a(g'(r)\frac{q}r|u|^2)_x+\ve q_{xx},\qquad \ \  b_t-(\vph(r)+\a\frac{g'(r)}r|u|^2+\ve q_x)b_x=0, \quad\  q=\sqrt{v_3^2+v_4^2}\label{e1.7}
\end{align}
We choose $g(r)$ with compact support in the interior of the physically relevant region $r\ge 0$ so as to preserve invariant domains in this region which contain the support of $g(r)$. In this way, our a priori estimates plus the compactness result in \cite{Se2} allow us to obtain the convergence of the solutions of the Cauchy problem for \eqref{e1.5},\eqref{e1.6},\eqref{e1.7}   as $\ve\to0$ and $\a=O(\ve^{1/2})$. 

\subsection{General symmetric systems}\label{sS:3} For general symmetric systems of conservation laws, that is, when the vector flux-function is the gradient of a real-valued function  $\vph(v)$, of a vector variable $v\in\R^n$, we propose the following model for SW-LW interactions
\begin{align} 
& iu_t+u_{xx}=u|u|^2+\a g(v)u,\label{e1.8}\\
&v_t+(\nabla_v\vph(v))_x=\a(\nabla_vg(v)|u|^2)_x+\ve v_{xx}.\label{e1.9}
\end{align}
As concrete examples, we consider the quadratic systems thoroughly analyzed in \cite{CK1,CK2} (see also, e.g., \cite{FS1,FS2} for other examples). For systems possessing bounded invariant domains, as the quadratic systems analized
in \cite{CK1,CK2}, we may suitably choose $g(v)$ with support contained in the interior of the invariant domain, so that
the compactness result obtained in those papers together with our a priori estimates lead to the strong convergence of
the solutions of the Cauchy problem for \eqref{e1.8},\eqref{e1.9}  as $\ve\to0$ and $\a=O(\ve^{1/2})$.

We remark that the case of the $p$-system of isentropic gas dynamics will be addressed elsewhere since it involves a longer discussion concerning the local uniform boundedness of  the solutions of the vanishing viscosity and interaction coefficient systems. For a basic account on the theory of conservation laws we refer to the recent books \cite{Br,Da,HR,LF,Se1}. The facts about the Schrödinger equation which will be used here belong to the very basic theory that can be found in general books of PDE such as, e.g., \cite{Ra}.
 
\subsection{Containts of the remaining sections} The remaining of this paper is organized according to the just made exposition. So, in Section~\ref{S:1} we briefly consider the simple case of scalar conservation laws.
We prove the convergence of the vanishing viscosity solutions with a fixed interaction parameter $\a$, sufficiently small. In Section~\ref{S:2}
we complement the study of the scalar case with the consideration of the degenerate symmetric system of \ref{sS:1}.
In Section~\ref{S:3} we consider SW-LW interactions for the $p$-system of the nonlinear elasticity with the special interaction function $g(v)=v$. Also here we prove the convergence of the vanishing viscosity solutions with a fixed interaction parameter $\a=1$. In the remaining three sections we consider the convergence of the solutions of the Cauchy problem for the SW-LW interactions for nonlinear systems of conservation laws with vanishing viscosity and interaction parameters. So, in Section~\ref{S:4} we consider again the SW-LW interactions for the $p$-system of nonlinear elasticity, now with a general interaction function $g(v)$. We prove the convergence of the solutions when $\ve\to0$ and $\a=O(\ve^{1/2})$. 
In Section~\ref{S:5}, we address the corresponding discussion for the system from nonlinear electromagnetism of \ref{sS:2}. 
The general symmetric system  and the particular example of the quadratic systems are considered in Section~\ref{S:6}.


\section{SW-LW interactions with scalar conservation laws}\label{S:1}

We consider the  system \eqref{e1.1},\eqref{e1.2}, which we repeat here for convenience 
\begin{align}
& i u_t + u_{xx} = |u|^2 u +  \a g(v)\, u  \label{e2.1} \\
& v_t + f (v)_x = \a(g'(v)\,|u|^2)_x +\ve v_{xx}, \label{e2.2}  
\end{align}
$x\in\R$, $t>0$. Here and henceforth the dependent variable $u(x, t)$ will always assume complex values, that is $u(x,t)\in\C$, while all other dependent variables will assume real values. In particular, $v(x, t)\in\R$. 

\begin{remark}\label{R:2.1} We will not distinguish the notation between the complex valued and the corresponding real valued  function spaces of the dependent variables, with the agreement that except for $u$ all others that appear in this paper ($v,r,w,p,q$, etc.) will be real valued (either scalar or vector, which will be clear in each context). 
\end{remark}  

 On the onset, we assume that  $f,g\in C^3(\R)$ are real functions such that $f(0)=0$ and $g'$ has compact support. 
 Besides, to allow the application of the compactness result in \cite{T}, based on the compensated compactness theory
 of  Tartar and Murat~\cite{T,Mu}, we impose the following nonlinearity condition: 
 \begin{equation}\label{e2.NL}
 \text{$\forall \k>0$, the set $\{ s\, : \, f''(s)-\k g'''(s) \not = 0 \}$ is dense in ${\R}$}. 
\end{equation}

We will study the Cauchy 
problem formed by \eqref{e2.1},\eqref{e2.2} and the initial data 
\begin{equation}
(u(x,0),v(x,0))=(u_0(x),v_0(x)),\qquad\text{with } u_0,\, v_0 \in H^1({\R}). \label{e2.3} 
\end{equation}

\begin{remark}\label{R:2.2} For simplicity of notation, throughout this paper we will always take the initial data independent of $\ve$. However, as it will become clear from the proofs, all the convergence results remain valid with essencially the same proof if the initial data relative to the conservation laws with viscosity and interaction terms (i.e., $v_0$, in the present case) were taken depending on $\ve$, uniformily bounded in $L^2(\R)\cap L^\infty(\R)$,   and converging in $L_\loc^1(\R)$ to a function in $L^2(\R)\cap L^\infty(\R)$. In particular, for the present section, we could assume $v(x,0)=v_0^\ve(x)$, $v_0^\ve\in H^1(\R)$, $\|v_0^\ve\|_{L^2(\R)}+\|v_0^\ve\|_{L^\infty(\R)}\le M$, for some $M>0$ independent of $\ve$, and $v_0^\ve\to v_0$ in $L_\loc^1(\R)$, with $v_0\in L^\infty(\R)\cap L^2(\R)$.
\end{remark}
 

 Existence of a local in time smooth solution for the Cauchy problem \eqref{e2.1},\eqref{e2.2},\eqref{e2.3} is obtained in a standard way using Duhamel's principle and Banach fixed point theorem (see, e.g., \cite{DFig}). We briefly sketch the argument as follows. For $R,T>0$, let
$$
B_R^T:=\{w\in X_T:=C([0,T];H^1(\R))\,:\,\|w\|_{L^\infty(0,T;H^1(\R))}\le R\}.
$$
We take $R>\max\{\|u_0\|_{H^1(\R)},\ \|v_0\|_{H^1(\R)}\}$. Define the map $\Phi:B_R^T\X B_R^T\to X_T\X X_T$ which associates to each pair $(\tilde u,\tilde v)\in B_R^T\X B_R^T$ the solution $(u,v)$ of the Cauchy problem
\begin{equation}\label{e2.FP}
\begin{cases} i u_t + u_{xx} = |\tilde u|^2 \tilde u +  \a g(\tilde v)\, \tilde u,\\ 
              v_t -\ve v_{xx}= - f(\tilde v)_x+ \a(g'(\tilde v)\,|\tilde u|^2)_x,\\
              u(0)=u_0,\qquad v(0)=v_0.
              \end{cases}
              \end{equation}
Hence, by Duhamel's principle, we have
\begin{align*}
& u(t)=U(t)u_0+\int_0^t U(t-s)\left[ |\tilde u|^2 \tilde u +  \a g(\tilde v)\, \tilde u\right](s)\,ds,\\
& v(t)=S(t)v_0-\int_0^t S(t-s)\left[ f(\tilde v)_x+ \a(g'(\tilde v)\,|\tilde u|^2)_x\right](s)\,ds,
\end{align*}
where $U(t):L^2(\R)\to L^2(\R)$ is the continuous unitary propagator associated with the Schr\"odinger equation and $S(t):L^2(\R)\to L^2(\R)$ is the continuous (contractive) semigroup associated with the
heat equation. 
Now, as it is well known, $U(t)\left(H^1(\R)\right)= H^1(\R)$, while $S(t)\left(L^2(\R)\right)\subset H^1(\R)$ for all $t>0$.  Since, using the well known inclusion $H^1(\R)\subset L^\infty(\R)$, we have 
$$
\left[|\tilde u|^2 \tilde u +  \a g(\tilde v)\, \tilde u\right]\in  X_T,
$$
and
$$
\left[ f(\tilde v)_x+ \a(g'(\tilde v)\,|\tilde u|^2)_x\right]\in C([0,T];L^2(\R)),
$$
if $(\tilde u,\tilde v)\in X_T\X X_T$, we deduce that for $T>0$ sufficiently small $\Phi$ is a (strict) contraction 
over the complete metric space $B_R^T\X B_R^T$. Therefore, we can apply  Banach fixed point theorem to obtain a unique local in time solution $(u,v)$ of the Cauchy problem \eqref{e2.1},\eqref{e2.2},\eqref{e2.3}. 

To be able to extend the local in time solution, whose existence and uniqueness we have just reviewed, we need some a priori estimates which will be also fundamental for the proof of the convergence of the vanishing viscosity sequence,
to be established subsequently, since they will not depend on $\ve$. 

These a priori estimate will guarantee, in particular, that for any $T>0$, if $(u(t),v(t))\in C([0,T);H^1(\R))$ is a solution of \eqref{e2.1},\eqref{e2.2},\eqref{e2.3},  then $u(x,t),v(x,t)$ satisfy    
\begin{align}
&iu_t+u_{xx}=h_1,\label{e2.a}\\
&v_t-\ve v_{xx}=h_2,\label{e2.b}
\end{align}
with $h_1\in L^2([0,T];H^1(\R))$ and $h_2\in L^2([0,T];L^2(\R))$. Therefore,  applying Duhamel's formula to the equation for $u_x$ obtained from \eqref{e2.a} by taking the $x$ derivative, we obtain $u\in C([0,T];H^1(\R))$.
On the other hand, from a well known result concerning equation \eqref{e2.b} it follows that $v\in L^2([0,T];H^2(\R))$ and $v_t\in L^2([0,T];L^2(\R))$, which in turn, also by a well known interpolation argument implies $v\in C([0,T];H^1(\R))$  (see, e.g., \cite{LM}). We then conclude that $u(t),v(t)$ may be extended to functions in $C([0,\infty);H^1(\R))$ which uniquely solve the Cauchy problem \eqref{e2.1},\eqref{e2.2},\eqref{e2.3} in $C([0,\infty);H^1(\R))^2$. 

Let us denote by $\|\,\cdot\,\|_p$ the norm in $L^p(\R)$, $1\le p\le\infty$ and agree that $\|\,\cdot\,\|_\infty$ will also be used as the $\sup$ norm of functions of the dependent variables. In what follows $c$ will denote any constant depending only on the data of the problem; its value may change  from one occurrence to the next. 

The first a priori estimate is an uniform bound for $\|v\|_\infty$ which is obtained from a sort of maximum principle.

\begin{lemma}\label{L:2.0} If $(u, v)\in C([0,T];H^1(\R))$ is a solution of \eqref{e2.1},\eqref{e2.2},\eqref{e2.3}, then  
\begin{equation}
\| v(t) \| _{\infty} \leq M, \qquad\text{for all $t\in[0,T]$}  \label{e2.4} 
\end{equation}
for some $M > 0$ independent of $\varepsilon$ and $t$.
\end{lemma}

\begin{proof} 1. Let $M>0$ be such that $[-M,M]$ is a closed interval whose interior $(-M,M)$ contains the support of $g'$ and the image of $v_0$. 

2. First,  it suffices to prove \eqref{e2.4} for the smooth solutions of the Cauchy problem obtained from \eqref{e2.1},\eqref{e2.2},\eqref{e2.3} by replacing \eqref{e2.2} by
\begin{equation}\label{e2.2'}
v_t + f (v)_x = \a(g'(v)\,|u|^2)_x +\nu h(v)+\ve v_{xx}, 
\end{equation}
for  $\nu\in(0,1)$, for  $h\in C^3(\R)$ such that $h(-M)>0$, $h(0)=0$, $h(M)<0$ (say, $h(v)=-v$), and \eqref{e2.3}
by
\begin{equation}\label{e2.3'}
(u(x,0),v(x,0))=(u_0^\nu(x),v_0^\nu(x)),\qquad (u_0^\nu,v_0^\nu)\in C^1([0,T], H^3(\R)),\ v_0^\nu(x)\in(-M,M),\ x\in\R.
\end{equation} 

3. Indeed, an unique local in time solution for \eqref{e2.1},\eqref{e2.2'},\eqref{e2.3'},  $(u^\nu,v^\nu)\in C([0,T'];H^1(\R))$, for some $T'>0$ independent of $\nu$, is obtained by using a fixed point argument totally similar to the one described above for obtaining the local in time solution of \eqref{e2.1},\eqref{e2.2},\eqref{e2.3}. Now, using the fact that $H^k(\R)$ is an algebra, $U(t)\left(H^k(\R)\right)=H^k(\R)$ and $S(t)\left(H^{k-1}(\R)\right)\subset H^k(\R)$, with $k=3$, we conclude that $(u^\nu,v^\nu)\in C([0,T'];H^3(\R))\cap C^1([0,T'];H^1(\R))$.  This  is true because  we can approach the fixed point in $C([0,T'];H^1(\R))$ by a sequence uniformly bounded in $C([0,T'];H^3(\R))\cap C^1([0,T'];H^1(\R))$ . In particular, $(u^\nu,v^\nu)\in C^1([0,T'];C^2([-R,R]))$ for all $R>0$.

4. We may easily obtain bounds independent of $\nu$  for $v^\nu$ in $L^2([0,T'];H^2(\R))$, for $v_t^\nu$ in $L^2([0,T'];L^2(\R))$,  for $u^\nu$ in $L^2([0,T'];H^1(\R))$ and for $u_t^\nu$ in $L^2([0,T];H^{-1}(\R))$. Hence, when $\nu\to0$, if $(u_0^\nu,v_0^\nu)\to (u_0,v_0)$ in $H^1(\R)$,  $(u^\nu,v^\nu)$  converges in $L^2([0,T];L^2(\R))$ to a pair $(u_*,v_*)\in L^2([0,T];H^1(\R))$ which satisfies the Duhamel's formulas
\begin{align}
& u(t)=U(t)u_0+\int_0^t U(t-s)\left[ |u|^2  u +  \a g( v)\, u\right](s)\,ds,\label{e2.DH1}\\
& v(t)=S(t)v_0-\int_0^t S(t-s)\left[ f( v)_x+ \a(g'( v)\,| u|^2)_x\right](s)\,ds.\label{e2.DH2}
\end{align}
We then get that $(u_*,v_*)\in C([0,T'];H^1(\R))$ and, thus, it coincides with  $(u(t),v(t))$ by the uniqueness of the local in time solution of \eqref{e2.1},\eqref{e2.2},\eqref{e2.3} in $C([0,T'];H^1(\R))$ . If $\|v^\nu\|_{\infty}\le M$ for $0\le t\le T'$,  for all $\nu\in(0,1)$, it follows that $\|v(t)\|_{\infty}\le M$, $0\le t\le T'$.

5. We prove that  $v^\nu(x,t)\in [-M,M]$, for all $(x,t)\in\R\X[0,T']$ and $\nu\in(0,1)$.    Indeed, since the support of $g'$ is contained in $(-M,M)$ and $v_0^\nu(x)\in(-M,M)$, $x\in\R$, the assumption that $v^\nu$ assumes values not belonging to $[-M,M]$ implies that there is a smallest time $t_*$, with $0<t_*<T'$, such that $v(x_*,t_*)\in\{-M,M\}$, for some $x_*\in\R$. This leads  easily to a contradiction by standard  maximum principle arguments. 

6. By choosing $T'$ depending only on $\sup_{t\in[0,T]}\|(u(t),v(t)\|_{H^1(\R)}$ and since $NT'\le T<(N+1)T'$, for some $N\in\N\cup\{0\}$, we may repeat the above argument for the time intervals $[kT',(k+1)T']$, $k=0,\dots,N$,  to obtain $\|v(t)\|_{\infty}\le M$ for all $t\in[0,T]$ which is \eqref{e2.4}.

\end{proof}

 


 
We now establish three relations which will be decisive for the achievement of our main energy estimates.

\begin{lemma}\label{L:2.1}  Let $(u, v) \in \left(C ([0, T) ; H^1({\R}) )\right)^2$, $T > 0$, be a solution of the  Cauchy problem \eqref{e2.1},\eqref{e2.2},\eqref{e2.3}. Then, we have 
\begin{align} 
  & {\frac{d}{dt}} \int_{\R} |u|^2 dx = 0.      \label{e2.5}   \\
   &  \notag  \\
 & {\frac{d}{dt}}\, \left[  \int_{\R} |u_x|^2 dx +  
 {\frac12} \int_{\R} |u|^4 dx +  \a \int_{\R} g(v)\, |u|^2 dx  
   -    \int_{\R} F(v) dx \right] \label{e2.6}  \\
     & \qquad + {\varepsilon} \a \int_{\R} ( |u |^2)_x g'(v) v_x  dx +
 {\varepsilon}  \a\int_{\R}  |u |^2 g''(v)\,(v_x)^2 dx - {\varepsilon}  \int_{\R}  f'(v) (v_x)^2  dx = 0 \notag  \\
 & \qquad\qquad {\rm where}\quad F(v) = \int_0^v f(s)\,ds.  \notag \\
 &  {\frac{d}{dt}} \left[{\frac1{2}} \int_{\R} v^2 dx +
 {\rm Im} \int_{\R} u \bar u_x  dx \right] +  {\varepsilon}   \int_{\R} (v_x)^2 dx = 0.   \label{e2.7} 
 \end{align}
\end{lemma}

\begin{proof} 1. By approximating the initial data in $H^1(\R)$ by functions in $C_0^\infty(\R)$ and using a  limit argument as in Lemma~\ref{L:2.0}, we may assume that $(u(x,t),v(x,t))$ is as smooth as needed. For simplicity we will make $\a=1$.  Now, equation \eqref{e2.5} is obtained in a standard way: we multiply \eqref{e2.1} by $\bar u$ to obtain an equation $(E1)$; take the conjugate of $(E1)$ to obtain an equation $(E2)$; make $(E1)-(E2)$ to obtain 
\begin{equation}\label{e2.s}
 i(|u|^2)_t+(\bar u u_x-u\bar u_x)_x=0,
\end{equation}
which gives \eqref{e2.5} upon integrating in $\R$. 

2. Equation \eqref{e2.7} is obtained  by first multiplying \eqref{e2.2} by $v$, integrating in $\R$ and using integration by parts to obtain
\begin{equation}\label{e2.7'}
\frac{d}{dt}\frac12\int_\R v^2\,dx+\ve \int_{\R}(v_x)^2=-\int_{\R}g(v)_x|u|^2\,dx.
\end{equation}
On the other hand,  multiplying \eqref{e2.1} by $\bar u_x$ to obtain an equation $(E3)$, taking the conjugate of $(E3)$ to obtain an equation $(E4)$ and making $(E3)+(E4)$, after a trivial manipulation, we obtain
$$
-({\rm Im\,}[u\bar u_x])_t +(|u_x|^2)_x=(\frac12|u|^4)_x+ (g(v)|u|^2)_x- g(v)_x|u|^2,
$$
which upon integration in $\R$ gives
$$
\frac{d}{dt}{\rm Im} \int_{\R} u \bar u_x\,  dx =\int_\R g(v)_x|u|^2\,dx.
$$
We then apply the last equation in \eqref{e2.7'} to obtain \eqref{e2.7} ({\em cf.}\ \cite{DFig}).

3. As to \eqref{e2.6}, we begin by taking the $x$ derivative of \eqref{e2.1}, obtaining an equation $(E5)$, then  multiplying $(E5)$ by $\bar u_x$ to obtain an equation $(E6)$, then  taking the conjugate of $(E6)$ to obtain an equation $(E7)$ and next making $(E6)-(E7)$ obtaining, after some trivial manipulation,
\begin{equation*}
i(|u_x|^2)_t+(\bar u_x u_{xx}-u_x\bar u_{xx})_x= 
[(\bar u_xu-u_x\bar u)(|u|^2+g(v))]_x-(|u|^2+g(v))(\bar u_xu-u_x\bar u)_x,
\end{equation*}
which, by using \eqref{e2.s}, gives
\begin{equation}\label{e2.6'}
i(|u_x|^2)_t+(\bar u_x u_{xx}-u_x\bar u_{xx})_x= 
[(\bar u_xu-u_x\bar u)(|u|^2+g(v))]_x-i(\frac12|u|^4)_t-i(g(v)|u|^2)_t+ig(v)_t|u|^2.
\end{equation}
Integrating \eqref{e2.6'} in $\R$ we obtain
\begin{equation}\label{e2.6''}
\frac{d}{dt}\int_\R[|u_x|^2+\frac12|u|^4+g(v)|u|^2]\,dx=\int_\R g(v)_t|u|^2\,dx.
\end{equation} 
To compute $\int_\R g(v)_t|u|^2\,dx$, we multiply \eqref{e2.2} by $g'(v)|u|^2$, obtaining, after some trivial manipulation,
$$
g(v)_t|u|^2+(|u|^2g'(v)f(v))_x-(|u|^2g'(v))_xf(v)+\ve v_x)=(\frac12(|u|^2g'(v))^2)_x+\ve g'(v)|u|^2v_{xx},
$$
which, by using $(g'(v)|u|^2)_x=v_t+f(v)_x-\ve v_{xx}$ , gives
\begin{multline*}
g(v)_t|u|^2+(|u|^2g'(v)f(v))_x-(v_t+f(v)_x-\ve v_{xx})f(v)\\
=(\frac12(|u|^2g'(v))^2)_x+\ve(g'(v)|u|^2v_x)_x-\ve g''(v)(v_x)^2|u|^2-\ve g'(v)v_x(|u|^2)_x.
\end{multline*}
Integrating in $\R$, it follows
\begin{align*}
& \int_{\R} |u |^2 g(v)_t\, dx  =  \int_{\R} (v_t+f(v)_x-\ve v_{xx})f(v)\, dx \\
&\qquad -  \varepsilon  \int_{\R} (|u |^2)_x g'(v) v_x\, dx - \varepsilon  \int_{\R} |u |^2 g''(v) (v_x)^2\, dx \\
&={\frac{d}{dt}} \int_{\R} F(v)\, dx + {\varepsilon}  \int_{\R} (v_x)^2 f'(v) dx\\
&\qquad -  \varepsilon  \int_{\R} (|u |^2)_x g'(v) v_x\, dx - \varepsilon  \int_{\R} |u |^2 g''(v) (v_x)^2\, dx,  
\end{align*}
thus establishing \eqref{e2.6}.
\end{proof}




{}From \eqref{e2.5} and \eqref{e2.7} we deduce
\begin{equation}  
{\frac1{2}} \| v (t)\|_2^2  +  {\varepsilon} \int_0^t  \|v_x(\tau)\|_{2}^2 \,d \tau \leq
c \left (1 +  \| u_x(t)\|_{2}  \right), \label{e2.8}  
\end{equation}
with $c$ independent of $\varepsilon $ and $T$. Thus, by \eqref{e2.4}, \eqref{e2.6} and \eqref{e2.8} it follows
\begin{equation}  
\| u_x(t) \|_{2}^2 \leq c + c\, \varepsilon \a\int_0^t  \int_{\R} |u |\,|u_x|\,|g'(v)|\,
|v_x|\,dx\, d\tau +  c\, \varepsilon \a\int_0^t  \int_{\R}  |u |^2 | g''(v)| (v_x)^2
\,dx\, d\tau.   \label{e2.9} 
\end{equation}


\begin{lemma}\label{L:2.2}  Let $(u, v) \in \left(C ([0, T) ; H^1({\R}) )\right)^2$, $T > 0$, be a solution of the  Cauchy problem \eqref{e2.1},\eqref{e2.2},\eqref{e2.3}.
Then there exists $\a_0\in(0,1]$ such that , for $0<\a\le\a_0$, $0<\varepsilon \leq 1$,
\begin{equation}
\|u_x(t)\|_{2}  \leq h (t), \label{e2.10}
\end{equation}
where $h\in C^\infty ([0, +\infty)) $ is a positive function independent of $T$ and $\varepsilon$.
\end{lemma}

\begin{proof} Let $ q(t) = 1+ \| u_x(t) \|_{2}^2 $. From \eqref{e2.4}, \eqref{e2.5} and \eqref{e2.9}, 
 using $ \| u \|_{\infty}^2 \leq c\, \| u \|_{2}\| u_x \|_{2}$, we deduce
$$ 
q(t) \leq c + c\,\varepsilon \a \int_0^t \| u_x \|_{2}^{3/2} \| v_x \|_{2} \,d\tau +
c\,\varepsilon \a \int_0^t \| u_x \|_{2} \| v_x \|_{2}^2 \,d\tau .
$$
Therefore, 
\begin{equation}\label{e2th}
q(t)\le \theta (t) := c +  c\,\varepsilon \a \int_0^t q(\tau)^{3/4}  \| v_x(\tau)\|_{2} \,d\tau
+ c\,\varepsilon \a \int_0^t  q(\tau)^{1/2} \| v_x(\tau)\|_{2}^2 \,d\tau. 
\end{equation}
We will prove that if $\a$ if sufficiently small, then
\begin{equation}\label{e2k}
\theta(t)\le c2^{t+1},\qquad \text{for $t\ge 0$},
\end{equation}
where $c$ is the same constant apearing in the definition of $\theta$.
 
Indeed, from \eqref{e2th} we have
\begin{align*}
\theta(t)&\le c+c\ve\a\sup_{0\le \tau\le t}\theta(\tau)^{3/4}t^{1/2}\left(\int_0^t\|v_x(\tau)\|_2^2\,d\tau\right)^{1/2}
+c\,\ve\a\sup_{0\le \tau\le t}\theta(\tau)^{1/2}\int_0^t\|v_x(\tau)\|_2^2\,d\tau\\
&\le c+c\ve\a (t^{1/2}+1)\sup_{0\le \tau\le t}\theta(\tau),
\end{align*}
where we have used \eqref{e2.8}. We then get
$$
\sup_{0\le \tau\le t}\theta(\tau)\le c+c\ve\a (t^{1/2}+1)\sup_{0\le \tau\le t}\theta(\tau).
$$
Let $\a_0>0$ be given by $\a_0:=1/(4c+1)$. Thus, for $0<\a\le\a_0$, if $0\le t\le 1$, we obtain
$$
\theta(t)\le\sup_{0\le \tau\le t}\theta(\tau)\le 2c.
$$
Now, we prove by induction that 
\begin{equation}\label{e2n}
\theta(t)\le c2^{n}\qquad\text{if $(n-1)< t\le n$}.
\end{equation}
The assertion is true for $n=1$ as we have just seen. Assume it holds for $(n-2)<t\le (n-1)$. Let $(n-1)<t\le n$.
By the definition of $\theta(t)$, we have 
$$
\theta(t)=\theta(n-1)+ c\,\varepsilon \a \int_{(n-1)}^t q(\tau)^{3/4}  \| v_x(\tau)\|_{2} \,d\tau
+ c\,\varepsilon \a \int_{(n-1)}^t  q(\tau)^{1/2} \| v_x(\tau) \|_{2}^2 \,d\tau.
$$
By the same estimates made in the case $n=1$, we get
$$
\sup_{n-1\le \tau\le t}\theta(\tau)\le \theta(n-1)+c\ve\a ((t-(n-1))^{1/2}+1)\sup_{n-1\le \tau\le t}\theta(\tau).
$$
As before, for $0<\a\le\a_0$ we get
$$
\theta(t)\le\sup_{n-1\le \tau\le t}\theta(\tau)\le 2\theta(n-1)\le c2^{n},
$$
by the induction hypothesis, which concludes the proof by induction of \eqref{e2n}, which in turn gives \eqref{e2k}.
 

\end{proof}


Finally, using the  estimates obtained in Lemmas~\ref{L:2.0} and \ref{L:2.2} we next establish the existence and uniqueness of global solutions $(u^\ve,v^\ve)$ of \eqref{e2.1},\eqref{e2.2},\eqref{e2.3}, $0<\ve\le 1$, and their convergence as $\ve\to0$ to a weak solution of 
\begin{align}
& i u_t + u_{xx} = |u|^2 u +  \a g(v)\, u ,  \label{e2.1B} \\
& v_t + f (v)_x = \a(g'(v)\,|u|^2)_x , \label{e2.2B}\\
&(u(x,0),v(x,0))=(u_0(x),v_0(x)).\label{e2.3B}  
\end{align}
Before stating the result we introduce the definition of  entropy solution for system \eqref{e2.1B},\eqref{e2.2B},\eqref{e2.3B}.

\begin{definition} \label{D:2.1} We say that the pair $(u(x,t),v(x,t))\in L_\loc^\infty(\R\X[0,\infty))^2$ is an entropy solution for \eqref{e2.1B},\eqref{e2.2B},\eqref{e2.3B} if:
\begin{enumerate}
\item[(i)] $u\in L_\loc^\infty([0,\infty);H^1(\R))\cap C([0,\infty);L^2(\R))$ with $u(0)=u_0$ in $L^2(\R)$ and
\begin{equation}\label{e2.entr1}
\int_0^t\int_{\R}i\,u\,\vph_t+u_x\vph_x+(|u|^2u+\a g(v)u)\vph\,dx\,dt=0,\qquad \forall \vph\in C_0^\infty(\R\X(0,\infty));
\end{equation}

\item[(ii)] for any $\eta\in C^2(\R)$ convex and $q_1,q_2\in C^2(\R)$ satisfying $q_1'(v)=\eta'(v)f'(v)$, $q_2'(v)=\eta'(v)g''(v)$ we have
\begin{equation}\label{e2.entr2}
\int_0^\infty\int_{\R} \eta(v)\phi_t+(q_1(v)-\a|u|^2q_2(v))\phi_x+\a(\eta'(v)g'(v)-q_2(v))(|u|^2)_x\phi\,dx\,dt+
\int_{\R}\eta(v_0(x))\phi(x,0)\,dx\ge0,
\end{equation}
for all non-negative $\phi\in C_0^\infty(\R^2)$.
\end{enumerate}
\end{definition}  
 
\begin{theorem}\label{T:2.3}  For each $0<\ve\le 1$ and $0<\a\le \a_0$, with $\a_0$ given by Lemma~\ref{L:2.1},  there exists a unique global solution
 $ (u^\varepsilon, v^\varepsilon) \in C ([0, +\infty) ; H^1({\R}))^2$
of the Cauchy problem \eqref{e2.1},\eqref{e2.2},\eqref{e2.3}. This solution satisfies
\eqref{e2.4}, \eqref{e2.5}, \eqref{e2.6}, \eqref{e2.7}, \eqref{e2.8} and \eqref{e2.10}, 
uniformly in $\varepsilon$. Moreover, there exists a subsequence of $(u^\ve,v^\ve)$ converging in 
$L_{\loc}^1(\R\X[0,\infty))^2$ to a pair of functions $(u,v)$, with $u\in C([0,\infty);H^1(\R))$ and
$v\in L^\infty(\R\X[0,\infty))$, which is an entropy solution of the problem \eqref{e2.1B},\eqref{e2.2B},\eqref{e2.3B}.
\end{theorem}
\begin{proof} 1. As we have already explained, the possibility of extending the local in time solution of \eqref{e2.1},\eqref{e2.2},\eqref{e2.3}, $(u^\ve,v^\ve)\in C([0,T);H^1(\R))^2$, follows from the fact that the estimates
\eqref{e2.4}, \eqref{e2.8} and  \eqref{e2.10} imply that $u^\ve$ and $v^\ve$ satisfy \eqref{e2.a} and \eqref{e2.b}, respectively, with $h_1\in L^2([0,T];H^1(\R))$ and $h_2\in L^2([0,T];L^2(\R))$. Therefore, we can extend $(u^\ve,v^\ve)$
to a unique global solution of \eqref{e2.1},\eqref{e2.2},\eqref{e2.3} in $C([0,\infty);H^1(\R))^2$.

2. The estimates \eqref{e2.4}, \eqref{e2.8} and  \eqref{e2.10} imply that $u^\ve$ is  bounded in $L^\infty([0,\infty);H^1(\R))$ and $v^\ve$ is  bounded in $L^\infty(\R\X[0,\infty))$ uniformly with respect to $\ve$.
A well known compactness lemma by Aubin (see, e.g., \cite{L}) applied to \eqref{e2.1} then implies the compactness of $u^\ve$ in $L_{\loc}^2([0,\infty);L^2(\R))$. 

3. As to $v^\ve$, the estimates \eqref{e2.4}, \eqref{e2.8} and  \eqref{e2.10} also imply that for any $\eta\in C^2(\R)$, if $q_1,q_2\in C^2(\R)$ are such that $q_1'(v)=\eta'(v)f'(v)$ and $q_2'(v)=\eta'(v)g''(v)$, then we have 
\begin{equation}\label{e2.comp}
\frac{\po}{\po t}\eta(v^\ve)+\frac{\po}{\po x}(q_1(v^\ve)-|u^\ve|^2q_2(v^\ve))\in\{\,\text{compact of }\,W_\loc^{-1,2}(\R\X(0,\infty))\,\}.
\end{equation}
Indeed, multiplying \eqref{e2.2} by $\eta'(v)$ we get, after simplification,
\begin{equation}\label{e2.comp2}
\frac{\po}{\po t}\eta(v)+\frac{\po}{\po x}(q_1(v)-|u|^2q_2(v))=(\eta'(v)g'(v)-q_2(v))(|u|^2)_x+\ve\frac{\po^2}{\po x^2} \eta(v)-\ve\eta''(v)(v_x)^2.
\end{equation}
Now, in the right-hand side of \eqref{e2.comp2} the terms $(\eta'(v)g'(v)-q_2(v))(|u|^2)_x-\ve\eta''(v)(v_x)^2$ are uniformly bounded in $L_\loc^1(\R\X(0,\infty))$, and hence compact in $W_\loc^{-1,q}(\R\X(0,\infty))$, for $1<q<2$, while the term $\ve\frac{\po^2}{\po x^2} \eta(v)$ is clearly compact in $W_\loc^{-1,2}(\R\X(0,\infty))$. Since the left-hand side of \eqref{e2.comp2} is uniformly bounded in $W_\loc^{-1,\infty}(\R\X(0,\infty))$, a well known interpolation argument in \cite{T} gives \eqref{e2.comp}.

4. Now \eqref{e2.comp} and the assumption \eqref{e2.NL} allow us to apply Tartar's method in \cite{T} to achieve the compactness of $\{v^\ve\}_{0<\ve\le1}$ in $L_\loc^1(\R\X[0,\infty))$, which together with the compactness of $\{u^\ve\}_{0<\ve\le 1}$ in the same space implies the existence of a subsequence of $(u^\ve,v^\ve)$ converging to a pair of functions $(u,v)$, with $u\in L_\loc^\infty([0,\infty);H^1(\R))$, $v\in L^\infty(\R\X[0,\infty))$,
 as $\ve\to0$.  
 
5.  We easily verify that $u$ satisfies Duhamel's formula \eqref{e2.DH1} and so $u\in C([0,\infty);L^2(\R))$ and $u(0)=u_0$ in $L^2(\R)$.  The verification of  \eqref{e2.entr1} is immediate while \eqref{e2.entr2} follows from \eqref{e2.comp2} by multiplying this equation by $\phi$, using integration by parts, and passing to the limit as $\ve\to0$. We then conclude that $(u,v)$ is an entropy solution of \eqref{e2.1B},\eqref{e2.2B},\eqref{e2.3B}. 
\end{proof}



\section{SW-LW interactions with a degenerate symmetric system}\label{S:2}


In this section,  we briefly consider the convergence of vanishing viscosity solutions of the system \eqref{e1.3},\eqref{e1.4} which couples Schrödinger equation with the degenerate symmetric system  introduced by Keyfitz and Kranzer \cite{KK}
with viscosity. For convenience we repeat the system here:
\begin{align}
& iu_t+u_{xx}=u|u|^2+\a g(r)u,\label{e3.1}\\
& v_t+(\vph(r)v)_x=\a (|u|^2g'(r)\frac{v}r)_x +\ve(r_{x}\frac{v}r)_x,\qquad r=|v|. \label{e3.2} 
\end{align}
The reason for including this system and considering it just after the SW-LW interactions with scalar conservation laws
is first to emphasize our criterion for suggesting a SW-LW coupling  motivated by the structure of the system governing the long waves, and second because the analysis pratically reduces to the scalar case. Indeed, the system \eqref{e3.2}, 
for the vector dependent variable $v\in\R^n$, may be split into a scalar conservation law with viscosity for $r$
\begin{equation}\label{e3.2'}
r_t+(\vph(r)r)_x=\a (|u|^2g'(r))_x +\ve r_{xx},
\end{equation}
and  transport equations for the components of $\varpi:=v/r$ 
\begin{equation}\label{e3.2''}
r\varpi_t+(r\vph(r)-\a|u|^2g'(r)-\ve r_x)\varpi_x=0.
\end{equation}  
We prescribe initial data for the system \eqref{e3.1},\eqref{e3.2}
\begin{equation}\label{e3.3}
(u(x,0),v(x,0))=(u_0(x),v_0(x)),
\end{equation}
and  assume that $v_0(x)=r_0(x)\varpi_0(x)$ with  $|\varpi_0(x)|=1$, $r_0(x)>0$ for $x\in\R$, and, for certain $r_*>0$,
we have
\begin{equation}\label{e3.4}
u_0\in H^1(\R),\qquad r_0-r_*\in H^2(\R)\quad\text{and}\quad\varpi_0\in L^\infty(\R).
\end{equation}

\begin{remark}\label{R:3.1} As in Remark~\ref{R:2.2} we could replace $v_0$ by $v_0^\ve$ with $v_0^\ve(x)=r_0^\ve(x)\varpi_0(x)$, ask that $r_0^\ve-r_*^\ve\in H^2(\R)$, $r_*^\ve>0$, $r_0^\ve(x)>0$ for $x\in\R$,  and impose the convergences 
$r_0^\ve\to r_0$ in $L_\loc^1(\R)$, with $r_0\in L^\infty(\R)$, $r_0(x)>\d_L>0$ for $x\in[-L,L]$, for all $L>0$, and $r_*^\ve\to0$, without changing the conclusions in this section.
\end{remark}   

As to $\vph$ and $g$ we assume $\vph,g\in C^3([0,\infty))$ and we also impose the following analogue of \eqref{e2.NL} 
\begin{equation}\label{e3.NL}
 \text{$\forall \k>0$, the set $\{ s\, : \, (r\phi(r))''-\k g'''(r) \not = 0 \}$ is dense in $[0,\infty)$}. 
\end{equation}

For the sake of reference let us write down separately the initial data for the system formed by \eqref{e3.1},\eqref{e3.2'}:
\begin{equation}\label{e3.3'}
(u(x,0),r(x,0))=(u_0(x),r_0(x))\quad\text{for $x\in\R$, $(u_0,r_0)\in H^1(\R)\X H^2(\R)$}.
\end{equation} 

A global and unique solution for \eqref{e3.1},\eqref{e3.2'},\eqref{e3.3'}, $(u,r)$ with $(u,r-r_*)\in C([0,\infty);H^1(\R))^2$, is obtained by a trivial adaptation of the analysis developed in the last section concerning the problem \eqref{e2.1},\eqref{e2.2},\eqref{e2.3}. Here we have an extra regularity for $r(x,t)$, namely $r-r_*\in C([0,\infty);H^2(\R))$ which easily follows from the assumption that $r_0-r_*\in H^2(\R)$. In particular, Lemmas~\ref{L:2.0}, \ref{L:2.1} and \ref{L:2.2} can easily be adapted to provide analogous results for the solution of the problem \eqref{e3.1},\eqref{e3.2'},\eqref{e3.3'}. More specifically, the arguments in Lemma~\ref{L:2.0} may easily be adapted to give the proof that for certain $r_{**}>0$, $r(x,t)\in [r_{**},M]$, where $r_{**}, M$ are such that the support of $g'(r)$ and the values of 
$r_0$ are contained in the open interval $(r_{**},M)$. Also, in Lemma~\ref{L:2.1} relations \eqref{e2.5} and \eqref{e2.6} have trivial analogues with $v$ replaced by $r$ and $f(v)$ replaced by $r\vph(r)$, while in relation \eqref{e2.7} we just need to replace $v$ by $r-r_*$ to obtain an analogue for the solution of the problem \eqref{e3.1},\eqref{e3.2'},\eqref{e3.3'}.  

Next we introduce the definition of renormalized solution of problem \eqref{e3.1},\eqref{e3.2},\eqref{e3.3}.

\begin{definition}\label{D:3.1} The pair $(u,v)\in C([0,\infty);L_{\loc}^1(\R))$ is a renormalized solution of \eqref{e3.1},\eqref{e3.2},\eqref{e3.3} with \eqref{e3.4} if  $(u(0),v(0))=(u_0,v_0)$ a.e.\ in $\R$,  $u\in C([0,\infty);H^1(\R))$, $v(x,t)=r(x,t)\varpi(x,t)$ where $r(x,t)>0$  for $(x,t)\in\R\X[0,\infty)$ and 
$r-r_*\in C([0,\infty);H^2(\R))$, $\varpi\in C([0,\infty);L^\infty(\R))$, $|\varphi(x,t)|=1$ a.e.\ in $\R\X[0,\infty)$,
and such that $(u,r)$ is the unique solution of \eqref{e3.1},\eqref{e3.2'},\eqref{e3.3'} in $C([0,\infty);H^1(\R))^2$
and $\varpi$  is a weak  solution of \eqref{e3.2''} in the sense that for all  $\phi\in C_0^\infty(\R\X(0,\infty))$ we have
\begin{equation}\label{e3.5}
\int_0^\infty\int_\R r\varpi\phi_t+(r\vph(r)-\a|u|^2g'(r)-\ve r_x)\varpi\phi_x\,dx\,dt=0.
\end{equation}
\end{definition}

Existence of a renormalized solution of \eqref{e3.1},\eqref{e3.2},\eqref{e3.3} with \eqref{e3.4} is obtained from the unique solution of \eqref{e3.1},\eqref{e3.2'},\eqref{e3.3'}, $(u,r)$ with $u\in C([0,\infty);H^1(\R))$ and $r-r_*\in C([0,\infty);H^2(\R))$, discussed above, in the following standard way. We defined a bi-Lipschitz change of coordinates $(x,t)\mapsto (y,t)$ where $y(x,t)$ is the solution of
$$
\begin{cases}\frac{\po y}{\po x}=r,\\ \frac{\po y}{\po t}= -r\vph(r)+\a|u|^2g'(r)+\ve r_x,\\ y(x,0)=\int_0^xr_0(z)\,dz.
\end{cases}
$$
We then define 
\begin{equation}\label{e3.6}
\varpi(x,t):=\varpi_0(y(x,t)).
\end{equation}
It is then easy to verify that $\varpi(x,t)$ so defined satisfies \eqref{e3.5} for all $\phi\in C_0^\infty(\R\X(0,\infty))$ and also that $\varpi\in C([0,\infty);L^\infty(\R))$ and $|\varphi(x,t)|=1$ a.e.\ in $\R\X[0,\infty)$. Hence, defining $v(x,t):=r(x,t)\varpi(x,t)$, we obtain that $(u(x,t),v(x,t))$ is a renormalized solution of \eqref{e3.1},\eqref{e3.2},\eqref{e3.3}.  Uniqueness of the renormalized solution is easily achieved by using the just mentioned change of coordinates $(x,t)\mapsto (y,t)$ in \eqref{e3.5} and taking suitable test functions to conclude that \eqref{e3.6} must hold a.e.\ in $\R\X[0,\infty)$. 

We are about to establish the convergence of a subsequence of the renormalized solutions of \eqref{e3.1},\eqref{e3.2},\eqref{e3.3}, $(u^\ve,v^\ve)$, as $\ve\to0$, to a weak solution of
 \begin{align}
& iu_t+u_{xx}=u|u|^2+\a g(r)u,\label{e3.1_0}\\
& v_t+(\vph(r)v)_x=\a (|u|^2g'(r)\frac{v}r)_x ,\qquad r=|v|. \label{e3.2_0} 
\end{align}
This weak solution $(u(x,t),v(x,t))$ will be such that $v(x,t)=r(x,t)\varpi(x,t)$ where $r(x,t)$ is an entropy solution of
\begin{equation}\label{e3.2'_0}
r_t+(\vph(r)r)_x=\a (|u|^2g'(r))_x,
\end{equation}
and $\varpi(x,t)$ is a weak solution of 
\begin{equation}\label{e3.2''_0}
r\varpi_t+(r\vph(r)-\a|u|^2g'(r))\varpi_x=0.
\end{equation} 
We make this assertion more precise with the following definition. 

\begin{definition} \label{D:3.2} We say that the pair $(u(x,t),v(x,t))\in L_\loc^\infty(\R\X[0,\infty))^2$ is a renormalized entropy solution for \eqref{e3.1_0},\eqref{e3.2_0},\eqref{e3.3} if:
\begin{enumerate}
\item[(i)] $v(x,t)=r(x,t)\varpi(x,t)$ with $r,\varpi\in L^\infty(\R\X[0,\infty))$, $r(x,t)>0$ and $|\varpi(x,t)|=1$ for almost all $(x,t)\in\R\X[0,\infty)$;

\item[(ii)] $(u,r)$ is an entropy solution of \eqref{e3.1_0},\eqref{e3.2'_0},\eqref{e3.3'} in the sense of Definition~\ref{D:2.1};

\item[(iii)] $\varpi$  is a renormalized   solution of \eqref{e3.2''_0} in the sense that for all $H\in C(\R^n)$ and  $\phi\in C_0^\infty(\R\X(0,\infty))$ we have
\begin{equation}\label{e3.7}
\int_0^\infty\int_\R rH(\varpi)\phi_t+(r\vph(r)-\a|u|^2g'(r))H(\varpi)\phi_x\,dx\,dt=0.
\end{equation}
\end{enumerate}

\end{definition}  


Concerning the convergence of the  renormalized solutions $(u^\ve,v^\ve)$ of \eqref{e3.1},\eqref{e3.2},\eqref{e3.3} when
$\ve\to0$ we have the following result.

\begin{theorem}\label{T: 3.3}  There exists $\a_0\in(0,1]$ such that for each $0<\ve\le 1$ and $0<\a\le \a_0$,   there exists a unique  global renormalized solution $(u^\ve,v^\ve)$ 
of the Cauchy problem \eqref{e3.1},\eqref{e3.2},\eqref{e3.3}.  Moreover, there exists a subsequence of $(u^\ve,v^\ve)$ converging in 
$L_{\loc}^1(\R\X[0,\infty))^2$ to a pair of functions $(u,v)$, with $u\in C([0,\infty);H^1(\R))$ and
$v\in L^\infty(\R\X[0,\infty))$,  which is a renormalized entropy solution of the problem \eqref{e3.1_0},\eqref{e3.2_0},\eqref{e3.3}.
\end{theorem}

\begin{proof} The existence of a unique global renormalized solution $(u^\ve,v^\ve)$ of \eqref{e3.1},\eqref{e3.2},\eqref{e3.3}  was explained above. For the compactness of $(u^\ve,r^\ve)$ we apply the Theorem~\ref{T:2.3}. The fact that $\varpi^\ve$ is also compact in $L_\loc^1(\R\X(0,\infty))$ follows by a well known argument of D.~Serre (see, e.g., \cite{Se1})  which may be explained as follows.
Since 
$\varpi^\ve$ is uniformly bounded in $L^\infty(\R\X(0,\infty))$, by the classical result on the existence of Young measures (see, e.g., \cite{T}), there is a subsequence still denoted by $\varpi^\ve$ and a Young measure $\mu_{x,t}$ such for all $H\in C(\R^n)$, the sequence $H(\varpi^\ve)$ weak-* converges to $\la\mu_{x,t}, H(\,\cdot\,)\ra$. Now, for any $H\in C^2(\R^n)$ and $\phi\in C_0^\infty(\R^2)$, multiplying by $H'(\varpi^\ve)\phi$ \eqref{e3.2''} (with $\varpi$ replaced by $\varpi^\ve$), integrating in $\R\X (0,\infty)$, using integration by parts,  and making $\ve\to0$ we obtain that
\begin{equation}\label{e3.T1}
\int_0^\infty\int_{\R} r\la\mu_{x,t},H(\,\cdot\,)\ra\phi_t+(r\vph(r)-\a g'(r)|u|^2)\la\mu_{x,t},H(\,\cdot\,)\ra\phi_t\,dx\,dt+\int_{\R}r_0(x)H(\varpi_0(x))\phi(x,0)\,dx =0.
\end{equation} 
A standard approximation argument extends this relation to all $H\in C(\R^n)$.
We then apply to \eqref{e3.T1} the change of coordinates $(x,t)\mapsto(y,t)$ with $y$ given by
$$
\begin{cases}\frac{\po y}{\po x}=r,\\ \frac{\po y}{\po t}= -r\vph(r)+\a|u|^2g'(r),\\ y(x,0)=\int_0^xr_0(z)\,dz,
\end{cases}
$$
to get that $\la\mu_{y,t},H(\,\cdot\,)\ra=H(\varpi_0(y))$ and so we obtain the strong convergence of $\varpi^\ve$ to a 
renormalized solution of \eqref{e3.2''_0} as defined in Definition~\ref{D:3.2}, which concludes the proof.

\end{proof}



\section{SW-LW interactions with the $p$-system of the nonlinear elasticity (I)}\label{S:3}  

Here we consider the system \eqref{e1.10}-\eqref{e1.12} in the special case where $g(v)=v$. Since in this case it will not be needed any restriction on $\alpha$ we just set $\alpha=1$. So we are going to study the Cauchy problem for the system
\begin{align} 
& iu_t+u_{xx}=u|u|^2+ vu,\label{e6.1}\\
& w_t-\s(v)_x=(|u|^2)_x +\ve w_{xx},\label{e6.2}\\
&v_t-w_x=0,\label{e6.3}
\end{align}    
with initial data
\begin{equation}\label{e6.4}
(u(x,0),w(x,0),v(x,0))=(u_0(x),w_0(x),v_0(x)),\qquad x\in\R,
\end{equation}
 satisfying
\begin{equation} 
u_0, \, w_0,\, v_0 \in H^1({\R})   \label{e6.4B}
\end{equation}
Following \cite{SeSh}, we define $\Sigma (v) = \int_0^v \sigma (s) ds$ ( we may assume $\sigma(0)=0$)
 and we impose the same conditions on the stress function $\sigma \in C^3({\R})$:

\begin{enumerate}

\item[(H1)]  $\sigma'(v) \geq \sigma_0 > 0$  with $\sigma_0$  constant. 
  
\item[(H2)] $\sigma'' (\lambda_0) = 0$ and  $\sigma''(\lambda) \not= 0$ for $\lambda \not = \lambda_0$. 
  
\item[(H3)] $\displaystyle  {\frac{\sigma''}{(\sigma')^{5/4}}}\,\,,
 \,\,\frac{\sigma'''}{(\sigma')^{7/4}}
\in L^2 ({\R}),\quad  \frac{\sigma''}{(\sigma')^{3/2}}\,\,, \,\,\frac{\sigma'''}{(\sigma')^2}
\in L^\infty ({\R})$. 

\item[(H4)]   $\displaystyle {\frac{\sigma(v)}{\Sigma(v)}} \longrightarrow 0 $ 
 as $|v|\rightarrow \infty$
and there exists constants $c > 0$ and $q > 1/2$ such that $(\sigma'(v))^q \leq c (1 + \Sigma (v))$.

\end{enumerate}

  In particular we have $ \Sigma (v) \geq  (\sigma_0/ 2) v^2$. To simplify we also suppose
that $v\in H^1({\R})$ implies $\int_{\R} \Sigma (v) dx < +\infty$. A typical example is given by $\s(v)=v^3+v$.

 Since the questions of global existence, uniqueness and corresponding a priori estimates for solutions of the problem
 \eqref{e6.1}-\eqref{e6.4} are addressed in the same way for general interaction functions $g(v)$ we will establish these results in the more general case of system \eqref{e1.10}-\eqref{e1.12}, which we repeat now for convenience
 \begin{align} 
& iu_t+u_{xx}=u|u|^2+ \a g(v)u,\label{e6.5}\\
& w_t-\s(v)_x=\a(g'(v)|u|^2)_x +\ve w_{xx},\label{e6.6}\\
&v_t-w_x=0.\label{e6.7}
\end{align}
and assume $g\in C^3(\R)$ with $g',g''$  uniformly bounded in $\R$.     
 
 
We start by proving the following lemma, analogue of Lemma~\ref{L:2.1}, for local in time smooth solutions of \eqref{e6.5},\eqref{e6.6},\eqref{e6.7},\eqref{e6.4}. 

\begin{lemma}\label{L:6.1}  Let $(u, v, w)\in C^1([0,  T) ; H^1({\R}))^3$, $T > 0$,
be a solution to the Cauchy problem \eqref{e6.5},\eqref{e6.6},\eqref{e6.7},\eqref{e6.4} in the interval $[0, T)$.
We have
\begin{align}
  & {\frac{d}{d t}} \int_{\R} |u|^2 dx = 0      \label{e6.8}   \\
   &   \notag  \\
 & {\frac{d}{d t}}\, \Bigl[  \int_{\R} |u_x|^2 dx +  
 {\frac12} \int_{\R} |u|^4 dx +   \a \int_{\R} g(v)\, |u|^2 dx   \label{e6.9}  \\
     & \qquad+ {\frac1{2}}  \int_{\R} w^2 dx +  \int_{\R} \Sigma(v) dx
\Bigr] + \varepsilon   \int_{\R} (w_x)^2 dx = 0 \notag   
\end{align}
Besides, if $\a\|g''\|_\infty$ is smaller than certain constant $\a_0>0$ depending only on the data (which is always true when $g(v)=v$) we have
\begin{align}
 & \frac{\ve\s_0}2 \int_0^t \int_{\R}  (v_x)^2 dx \,d\tau + \varepsilon^2
 \int_{\R} (v_x)^2 dx \leq c_0 (1 + t),   \label{e6.10}
\end{align}
for $\varepsilon \leq 1$, where $c_0 > 0$ is a constant independent of $\varepsilon$ and $T$.
\end{lemma}

\begin{proof} 1. The proof of \eqref{e6.8} is identical as the corresponding relation in Lemma~\ref{L:2.1}.

2. Concerning \eqref{e6.6}, as for the corresponding relation in Lemma~\ref{L:2.1}, it is enough to compute 
$\int_{\R} |u|^2 g(v)_t\, dx$, assuming without loss of generality $\a=1$ and enough regularity for the solution. Now, to do that we multiply \eqref{e6.7} by $|u|^2g'(v)$ and use \eqref{e6.6} to obtain
$$
|u|^2g(v)_t=(|u|^2g'(v)w)_x-(|u|^2g'(v))_xw=(|u|^2g'(v)w)_x-(w_t-\s(v)_x-\ve w_{xx})w.
$$
Integrating in $\R$ and using integration by parts we then get
\begin{align*}
& \int_{\R} |u|^2g(v)_t \,dx= - \int_{\R} (w_t - \sigma(v)_x) w \,dx + 
  \varepsilon  \int_{\R} w w_{xx} \,dx \\
 = & -{\frac1{2}}\,  {\frac{d}{dt}} \int_{\R} w^2 \,dx - 
\int_{\R} \sigma(v) v_t \,dx -  \varepsilon \int_{\R}  (w_{x})^2 \,dx  \\
 = &   -{\frac1{2 }} \, {\frac{d}{d t}} \int_{\R} w^2 \,dx 
 - \, {\frac{d}{dt}} \int_{\R} \Sigma(v)\, dx - 
{\varepsilon}  \int_{\R}  (w_{x})^2 \,dx. 
\end{align*}

 3. To prove \eqref{e6.7} we follow closely the proof of Lemma~1, (ii) in \cite{SeSh}.  We multiply \eqref{e6.6} by $v_x$, integrate the resulting equation in $\R$, use \eqref{e6.7} and integration by parts, to obtain
\begin{align*}
&  \int_{\R} (w_t v_x - \sigma'(v) (v_x)^2 ) \,dx =  \a\int_{\R} (g'(v)|u|^2)_x v_x\,dx +
  {\frac\varepsilon2}  \int_{\R}  {\frac{\partial}{\partial t}} (v_x)^2 dx.
\end{align*}
Now, observing that $w_tv_x=(wv_x)_t-ww_{xx}$ we obtain, after integration by parts,
\begin{align*}
 &     {\frac{d}{d t}} \int_{\R}(w_x v)\,dx +  \int_{\R} (w_x)^2 dx - 
   \int_{\R}  \sigma'(v) (v_x)^2  dx     \\  
    & \qquad =   \a\int_{\R} (g'(v)|u|^2)_x v_x\,dx +
  {\frac\varepsilon2}{\frac{d}{d t}}  \int_{\R}   (v_x)^2 dx.
\end{align*}
Integrating in $[0, t]$ we get
\begin{align*}
 \int_{\R} (w_x v)\,dx & -  \int_{\R} w_{0x} v_0 dx + \int_0^t  \int_{\R} (w_x)^2 dx\, d\tau 
 - \int_0^t  \int_{\R} \sigma'(v) (v_x)^2 dx\, d\tau \\
 & =  \a\int_0^t  \int_{\R}  (g'(v)|u|^2)_x v_x\,dx\, d\tau +  {\frac\varepsilon2}  \int_{\R}
 (v_x)^2 dx - {\frac\varepsilon2}  \int_{\R} (v_{0x})^2 dx, 
 \end{align*}
 and then
 \begin{align}
 & \int_0^t  \int_{\R} \sigma'(v) (v_x)^2 dx\, d\tau 
 + {\frac\varepsilon2}  \int_{\R} (v_x)^2 dx \,\leq\,   {\frac\varepsilon4} \int_{\R} (v_x)^2 dx
+ {\frac1\varepsilon}  \int_{\R} w^2 dx\label{e6.11} \\
& + \int_{\R} |w_{0x}|\,|v_0|\,dx + {\frac\varepsilon2}  \int_{\R} (v_{0x})^2 dx +
 \int_0^t  \int_{\R} (w_x)^2 dx\, d\tau +  \a \int_0^t  \int_{\R}|g''(v)||u|^2|v_x|^2+ 2|u|\,|u_x|\,|v_x|\,dx\, d\tau.\notag 
\end{align}
Moreover, by \eqref{e6.5} and \eqref{e6.6}, we have
\begin{align}
&\|u\|_\infty\le c,\label{e6.12}\\
&2  \int_0^t  \int_{\R} |u|\,|u_x|\,|v_x|\,dx\, d\tau \leq c\, t + {\frac{\sigma_0}4}
 \int_0^t  \int_{\R} (v_x)^2 dx\, d\tau, \label{e6.13} 
\end{align}
with $c$ independent of $\varepsilon$. By the  inequalities \eqref{e6.11}, \eqref{e6.12}, \eqref{e6.13} and 
\eqref{e6.5}, \eqref{e6.6} we deduce \eqref{e6.7} for $\a\|g''\|_\infty$ sufficiently small, depending only on the data, and $\varepsilon \leq 1$.
\end{proof}


 The well posedness of the Cauchy problem \eqref{e6.5},\eqref{e6.6},\eqref{e6.7},\eqref{e6.4} is established in the following result.

\begin{lemma}\label{L:6.2}  For each $\varepsilon \leq 1$ and $\a\|g''\|_\infty$ sufficiently small, depending only on the data, the  Cauchy problem \eqref{e6.5},\eqref{e6.6},\eqref{e6.7},\eqref{e6.4}
has a unique global solution $(u^\varepsilon, w^\varepsilon, v^\varepsilon) \in
C ([0, +\infty) ; H^1({\R}))^3$. 
\end{lemma}

\begin{proof} The proof is very similar to that of the existence part of Theorem~\ref{T:2.3}. Preserving the same notation, we consider the map
$$ 
(\tilde u, \tilde v) \longrightarrow \Phi  (\tilde u, \tilde v) = (u, v) \in X_T \times X_T,
$$
where $(u(x,t),v(x,t))$ is the  solution of the linear problem (we put $\varepsilon = 1$ to simplify)
\begin{align*}
& i u_t + u_{xx} = |\tilde u|^2 \tilde u + \a g(\tilde v) \tilde u  \\
&  v(t)  = v_0 + \int_0^t w_x(\tau) \,d\tau  
\end{align*}
where $w\in X_T$, satisfying $w\in L^2((0, T) ; H^2)$, $w_t\in L^2((0, T) ; L^2)$, is the 
unique solution of the linear initial value problem
\begin{align*}
&w_t - w_{xx} = \sigma (\tilde v)_x + \alpha (g'(\tilde v)|\tilde u|^2)_x,\\ 
&w (0) = w_0.
\end{align*}
Recall that, 
$$ 
u(t) = U(t) u_0 + \int_0^t U(t - s) [ \,|\tilde u|^2 \tilde u + \a g(\tilde v) \tilde u \,] (s)\,ds  
$$
and $(\sigma (\tilde v) +\alpha\, |\tilde u|^2 )_x \in C ([0, T] ; L^2)$. The existence of a local solution
is again an easy consequence of the Banach fixed point theorem for suitables $R$ and $T$.

  Moreover, since we have
$$
 w_t - w_{xx} = \sigma' (v) v_x +  \alpha g''(v)v_x|u|^2+ 2\a g'(v)\Re(u \bar u_x) , 
 $$
we deduce, from \eqref{e6.5}, \eqref{e6.6} and \eqref{e6.8}, an apriori estimate
$$ 
\| w_t - w_{xx} \|_{L^2((0, T) ; L^2(\R))} \leq c (T), \quad c\in C([0, +\infty))  
$$
and so $w\in L^2 ((0, T) ; H^2)$ with a similar estimate for  $\| w \|_{L^2((0, T) ; H^2(\R))}$.
It is now easy to obtain suitable apriori estimates for $u$, $w$ and $v$ in 
$C([0, T] ; H^1)$ and the proof is completed.  

\end{proof}


We now establish the convergence of the vanishing viscosity solutions of \eqref{e6.1}-\eqref{e6.4}. The main point in this special case where $g(v)=v$ is that the  term $(|u|^2)_x$ is independent of $v$ and by the Lemma~\ref{L:6.1}
it is bounded in $L_\loc^1(\R\X[0,\infty))$ uniformly with respect to $\ve$. Below,  by $v\in L_\loc^{\Sigma}(\R\X[0,\infty))$ we mean that $\int_K\Sigma(v)\,dx\,dt<+\infty$ for any compact set $K\subset\R\X[0,\infty)$.

\begin{theorem}\label{T:6.3} Let $(u^\ve,w^\ve,v^\ve)$ be the global solution of \eqref{e6.1}-\eqref{e6.4} given by Lemma~\ref{L:6.2}.  There exists a subsequence of $(u^\ve,w^\ve, v^\ve)$ converging in 
$L_{\loc}^1(\R\X[0,\infty))^3$ to a  function $(u,w,v)$, with $u\in C([0,\infty);H^1(\R))$,
$w\in L_\loc^2(\R\X[0,\infty))$ and $v\in L_\loc^{\Sigma}(\R\X[0,\infty))$, which is a weak solution of the problem 
\begin{align} 
& iu_t+u_{xx}=u|u|^2+ vu,\label{e6.1'}\\
& w_t-\s(v)_x=(|u|^2)_x,\label{e6.2'}\\
&v_t-w_x=0,\label{e6.3'}\\
&(u(x,0),w(x,0),v(x,0))=(u_0(x),w_0(x),v_0(x)),\qquad x\in\R.\label{e6.4'}
\end{align} 
\end{theorem}

\begin{proof} 1. As in the  proof of Theorem~\ref{T:2.3}, the compactness of $u^\ve$ is a consequence of the classical Aubin's lemma \cite{L}, since by Lemma~\ref{L:6.1}, and for each $T > 0$, we have
\begin{align*}
& \{  u^{\varepsilon} \}_\varepsilon \quad \hbox{\rm bounded in } \quad L^\infty ((0, T) ; H^1),   \\
& \{  u_t^{\varepsilon} \}_\varepsilon \quad \hbox{\rm bounded in } \quad L^\infty ((0, T) ; H^{-1}),   
\end{align*}
and so $ \{  u^{\varepsilon} \}_\varepsilon$ lies in a compact set of $L^2 ((0, T); L^2(I_R))$
for each interval $I_R = \{ x\in {\R} : | x | \leq R \},\,\,R > 0$. Hence, by applying a 
standard diagonalization procedure, we conclude that there exists 
$u\in L^\infty ({\R}_{+} ; H^1({\R}))$ and a subsequence of 
$ \{  u^{\varepsilon} \}_\varepsilon $, still denoted by $ \{  u^{\varepsilon} \}_\varepsilon$,
such that $u^\ve\to u$ in $L_\loc^1(\R\X(0,\infty))$.

2. Also by Lemma~\ref{L:6.1}, we have that $w^\ve$ is uniformly bounded in $L_\loc^2(\R\X(0,\infty))$, $v^\ve$ is
uniformly bounded in $L_\loc^{\Sigma}(\R\X(0,\infty))$ and for any entropy-entropy flux pair $(\eta,q)$ for \eqref{e6.2'},\eqref{e6.3'} with $\eta_w(w,v), \eta_v(w,v)$ uniformly bounded in $\R^2$ we have
\begin{equation}\label{e6.SeSh}
\eta(w^\ve,v^\ve)_t+q(w^\ve,v^\ve)_x\in \{\,\text{compact of }\,W_\loc^{-1,2}(\R\X(0,\infty))\,\}.
\end{equation}
Hence, we can use the compactness result in \cite{SeSh} (see also \cite{DF}) to conclude that $(w^\ve,v^\ve)$ is precompact in $L_\loc^1(\R\X(0,\infty))$ and also that any limit of a subsequence of $(u^\ve,w^\ve,v^\ve)$ is a weak solution of \eqref{e6.1'}-\eqref{e6.4'}.  

\end{proof}
 

\section{SW-LW interactions with the $p$-system of the nonlinear elasticity (II)}\label{S:4}  

It is somehow surprising that one is able to carry out the vanishing viscosity method for the solutions of \eqref{e6.1}-\eqref{e6.4} obtaining thus a weak solution of \eqref{e6.1'}-\eqref{e6.4'}, since SW-LW interactions
constitute a microscopic phenomenon while the limit as $\ve\to0$ is only physically meaningful at a macroscopic level.
This comment applies as well to the  two cases considered before, namely, the case of scalar conservation laws and that
of the symmetric degenerate systems, but one could argue that  these are less realistic models. 

In the specific case of the nonlinear elasticity  with physical viscosity, the method that we used to justify the vanishing viscosity limit is only able to handle the particular case $g(v)=v$, because, as we already mentioned, in this case the term $(g'(v)|u|^2)_x$ in \eqref{e6.6} is independent of $v$. In the more general case of the problem 
\eqref{e6.5}-\eqref{e6.8}, assuming only that $g'(v)$ and $g''(v)$ are bounded in $\R$, a similar procedure can justify the limit when both viscosity $\ve$ and interaction coefficient $\a$ go to zero, with $\a=O(\sqrt{\ve})$. This will be also the limit process to be studied in the remaining sections, which we believe to be a more realistic one.

Thus we next establish the convergence of the vanishing viscosity and interaction coefficient for the solutions of 
\eqref{e6.5}-\eqref{e6.8}.

\begin{theorem}\label{T:7.1} Let $(u^\ve,w^\ve,v^\ve)$ be the global solution of \eqref{e6.5}-\eqref{e6.8} given by Lemma~\ref{L:6.2} and assume $\a=O(\sqrt{\ve})$.  There exists a subsequence of $(u^\ve,w^\ve, v^\ve)$ converging in 
$L_{\loc}^1(\R\X[0,\infty))^3$ to a  function $(u,w,v)$, with $u\in C([0,\infty);H^1(\R))$,
$w\in L_\loc^2(\R\X[0,\infty))$ and $v\in L_\loc^{\Sigma}(\R\X[0,\infty))$, which is a weak solution of the problem 
\begin{align} 
& iu_t+u_{xx}=u|u|^2,\label{e7.1}\\
& w_t-\s(v)_x=0,\label{e7.2}\\
&v_t-w_x=0,\label{e7.3}\\
&(u(x,0),w(x,0),v(x,0))=(u_0(x),w_0(x),v_0(x)),\qquad x\in\R.\label{e7.4}
\end{align} 
\end{theorem}

\begin{proof} This follows as in the proof of Theorem~\ref{T:6.3}, since the fact that $\a=O(\sqrt{\ve})$ gives us that
the term $\a(g'(v)|u|^2)_x$ is again uniformly bounded in $L_\loc^1(\R\X(0,T))$. Thus, for any entropy-entropy flux pair $(\eta,q)$ for \eqref{e7.2},\eqref{e7.3} with $\eta_w(w,v), \eta_v(w,v)$ uniformly bounded in $\R^2$ we again have
\begin{equation}\label{e7.SeSh}
\eta(w^\ve,v^\ve)_t+q(w^\ve,v^\ve)_x\in \{\,\text{compact of }\,W_\loc^{-1,2}(\R\X(0,\infty))\,\},
\end{equation}
which allows us  to apply the compactness result in \cite{SeSh}.

\end{proof}
  

 

\section{SW-LW interactions with the equations of nonlinear electromagnetism}\label{S:5}


In this section we consider the convergence of the vanishing viscosity and interaction coefficient solutions of the Cauchy problem for the system \eqref{e1.5}-\eqref{e1.7} representing the interaction of short waves described by the nonlinear Schrödinger equation with  long waves modelling nonlinear electromagnetism. Convergence of the vanishing viscosity method for the system of  nonlinear eletromagnetism was proved in \cite{Se2} and we are going to apply the compactness result in \cite{Se2} to study the convergence of solutions of the Cauchy problem for \eqref{e1.5}-\eqref{e1.7} when $\ve,\a\to0$ with $\a=O(\ve^{1/2})$. 

Since the transport equations in \eqref{e1.6} and \eqref{e1.7} are treated exactly as in the case of the degenerate symmetric systems in Section~\ref{S:2}, here we just analyse the equations for $u,p,q$, that is,
\begin{align}
& iu_t+u_{xx}=u|u|^2+\a g(r)u, \qquad r=\sqrt{p^2+q^2}\label{e4.1}\\
& p_t+(\vph(r)p)_x=-\a(g'(r)\frac{p}r|u|^2)_x+\ve p_{xx}, \label{e4.2}\\
& q_t-(\vph(r)q)_x=\a(g'(r)\frac{q}r|u|^2)_x+\ve q_{xx}.\label{e4.3}
\end{align}
We suppose that we are given initial data
\begin{equation}\label{e4.4}
(u(x,0),p(x,0), q(x,0))=(u_0(x),p_0(x),q_0(x)).
\end{equation}
As in the case of the degenerate symmetric systems, we assume $p_0(x)>0$, $q_0(x)>0$, for $x\in\R$, and for certain $p_*>0$ and $q_*>0$ we have  
\begin{equation}  
u_0 \in H^1({\R})\ \text{and}\ p_0-p_*,q_0-q_*\in H^2(\R). \label{e4.5} 
\end{equation}
A remark similar to Remark~\ref{R:3.1} also holds here for both $p_0$ and $q_0$.

We also assume that $g\in C^3([0,\infty))$ and $g'$ has compact support contained in $(0,\infty)$, and  
 $ \varphi \in  C^2 ([0, +\infty) )$ is a real function such that 
 \begin{equation}\label{e4.5B}
 \varphi (r) \geq a > 0\ \text{and}\ \varphi'(r) \ge  0\ \text{for $r\in [0, +\infty)$}.
\end{equation}

Positively invariant regions in the quadrant $p\ge0,\,q\ge0$ for the system \eqref{e4.2},\eqref{e4.3} without the interaction terms, i.e.\ $\a=0$, were obtained in \cite{Se2}. When these terms are present as in the coupled system
\eqref{e4.1}-\eqref{e4.3}, positively invariant regions are obtained as in the case of the scalar conservation laws,
due to our assumption that the support of $g'$ is compact in $(0,\infty)$. More specifically, we have the following.

\begin{lemma}\label{L:4.1}  Let $(u, p, q)(x,t)$ be a smooth solution of \eqref{e4.1}-\eqref{e4.4} defined in $\R\X[0,T)$ with $u_0,p_0,q_0,g,\varphi$ satisfying the hypotheses above. Then there exists a constant $M_0 > 0$ such that
\begin{equation} 
0 \leq p(x, t) \leq M_0,\quad 0\le q(x,t)\le M_0,\quad x\in {\R}, \quad t\in [0, T),  \label{e4.5'}  
\end{equation}
with $c$ independent of $ \varepsilon$ and $T$.
\end{lemma}

\begin{proof} We recall that the positively invariant regions for the system \eqref{e4.2},\eqref{e4.3} with $\a=0$ obtained in \cite{Se2} are convex regions of the form $D=\{(p,q)\in[0,\infty)^2\,:\, 0\le p\le h(p),\ 0\le q\le h(q)\}$
where $h\in C^3([0,\infty))$ is decreasing with $h(0)>0$, $h(R)=0$ for some $R>0$, $h''(s)<0$ for $0\le s<s_*$, $h''(s)>0$ for $s_*<s\le R$, for certain $s_*\in[0,R]$ satisfying $h(s_*)=s_*$ and $h'(s_*)<-1$. Let the support of $g'(r)$ be contained in an interval $[m,M]$ with $0<m<M$, let $D$ be such an invariant domain obtained in \cite{Se2},
and assume that:
\begin{enumerate}
\item[(i)] the image of $(p_0,q_0)$ is contained in the interior of $D$;

\item[(ii)] $\{(p,q)\in[0,\infty)^2\,:\,m\le r=\sqrt{p^2+q^2}\le M\}\subset D$.
\end{enumerate}
As in the scalar case in Section~\ref{S:1}, we first consider the perturbed system
\begin{align}
& iu_t+u_{xx}=u|u|^2+\a g(r)u, \qquad r=\sqrt{p^2+q^2}\label{e4.1p}\\
& p_t+(\vph(r)p)_x=-\a(g'(r)\frac{p}r|u|^2)_x+\nu (p_\bullet-p)+\ve p_{xx}, \label{e4.2p}\\
& q_t-(\vph(r)q)_x=\a(g'(r)\frac{q}r|u|^2)_x+\nu (q_\bullet-q)+\ve q_{xx},\label{e4.3p}
\end{align}
where $(p_\bullet,q_\bullet)$ is any point in the interior of $D$. Because $D$ satisfies  (i) and (ii) above, and the terms $-\a(g'(r)\frac{p}r|u|^2)_x$ and  $\a(g'(r)\frac{q}r|u|^2)_x$ in \eqref{e4.2p} and \eqref{e4.3p}, respectively, vanish at $\po D$, assuming that there is a first time at which the solution leaves the region $D$ leads us to contradiction
exactly as in the case where $\a=0$ addressed in \cite{Se2}. A standard limit argument as $\nu\to0$ then gives the desired boundedness for the solution of \eqref{e4.1}-\eqref{e4.4}.     

\end{proof}


We now establish two identities similar to \eqref{e2.5} and \eqref{e2.6} in Lemma~\ref{L:2.1}.
 
\begin{lemma}\label{L:4.2}  Let $(u,p,q)$ with $(u, p-p_*, q-p_*) \in  C([0, T) ; H^1({\R}) )^3$, $T > 0$,
be a solution of the  Cauchy problem \eqref{e4.1}-\eqref{e4.4}. We have
\begin{align}
  & {\frac{\partial}{\partial t}} \int_{\R} |u|^2 dx = 0      \label{e4.6}   \\
   &  \notag  \\
 & {\frac{\partial}{\partial t}}\, \left[ {\frac12} \int_{\R} |u_x|^2 dx +  
 {\frac14} \int_{\R} |u|^4 dx +   {\frac{\alpha}2} \int_{\R} g(r)\, |u|^2 dx  
   + {\frac14}   \int_{\R} H(r) dx \right]   \label{e4.7}          \\
     & \qquad + {\frac\varepsilon2}  \int_{\R} \varphi(r)
\,(|p_x|^2 + |q_x|^2)  dx +
 {\frac\varepsilon{2 \alpha}}  \int_{\R}  \varphi'(r) r \,(r_x)^2 dx \,\notag  \\
 & \qquad + { \frac{\varepsilon\alpha} 2}  \int_{\R}  \left(|u|^2\right)_x g'(r) r_x dx 
+{ \frac{\varepsilon\alpha} 2}  \int_{\R} |u|^2\left( g''(r) - 
 {\frac{g'(r)} r} \right) (r_x)^2  dx = 0, \notag   
\end{align}
where $\displaystyle  H(r) = \int_0^r s \varphi(s)\,ds$.
\end{lemma}

\begin{proof} 1. The proof of \eqref{e4.6} is identical to that of \eqref{e2.5} so we omit it.

2. As to the proof of \eqref{e4.7}, it is very similar to that of \eqref{e2.6} and after proceeding as in the beginning  of that proof we arrive at the point where we need to compute $\int_{\R}g(r)_t|u|^2\,dx$. To do that, now we have to multiply equation \eqref{e4.2} by $g'(r)\dfrac{p}{r}|u|^2$ obtaining an equation $(E1)$, multiply equation \eqref{e4.3} by $g'(r)\dfrac{q}{r}|u|^2$ obtaining an equation $(E2)$, add up $(E1)+(E2)$ obtaining an equation $(E3)$ and integrate $(E3)$. The remaining of the proof follows as in the proof of \eqref{e2.6}. 
\end{proof}


We need one more relation which together with \eqref{e4.6} and \eqref{e4.7} will allow us to obtain the necessary energy estimates.


\begin{lemma}\label{L:4.3}  Under the hypothesis of the Lemma~\ref{L:4.2} we have
\begin{align}
\int_{\R} H(r) dx & + a\,\varepsilon \int_0^t \int_{\R} 
\,(|p_x|^2 + |q_x|^2) \, dx\,  d\tau  \label{e4.8}  \\
& \le c \left(1 + \alpha  \int_0^t \int_{\R} |u|^2 (|p_x| + |q_x|) dx\,  d\tau \right),    
\notag
\end{align}
with $c$  independent of $\varepsilon, T$ and $\alpha$. 
\end{lemma}


\begin{proof} We deduce from \eqref{e4.2}, \eqref{e4.3} 
\begin{align*}
& \int_{\R} p_t \varphi(r)\, p\, dx = -\alpha \int_{\R} 
\left( g'(r)\, \frac{p}{r}\, |u|^2\right)_x   \varphi(r)\, p \,dx + \varepsilon  \int_{\R}
 \varphi(r) p\, p_{xx}\, dx,   \\
    \\
& \int_{\R} q_t \varphi(r)\, q\, dx =\alpha \int_{\R} 
\left( g'(r)\, \frac{q}{r}\, |u|^2\right)_x   \varphi(r)\, q \,dx + \varepsilon  \int_{\R}
 \varphi(r) q\, q_{xx}\, dx,
 \end{align*}
 and so,
 \begin{align*}
 & {\frac{\partial}{\partial t}} \int_{\R} H(r) dx =  \int_{\R}  \varphi(r)\,r
\left(\frac{p}{ r}\,p_t + \frac{q}{ r}\,q_t\right) dx = \alpha  \int_{\R}
\left( g'(r) \,\frac{p}{ r}\,|u|^2\right) ( \varphi(r)\,p)_x dx -  \\
& \qquad - \varepsilon  \int_{\R}  \varphi'(r)\,p \left(\frac{p}{ r}\,p_x +
 \frac{q}{r}\,q_x\right) p_x dx - \varepsilon  \int_{\R}  \varphi(r) (p_x)^2 dx -  \\
& \qquad - \alpha  \int_{\R} \left( g'(r)\, \frac{p}{ r}\,|u|^2 \right) 
(\varphi(r) \,q)_x dx -   \varepsilon  \int_{\R}  \varphi'(r)\,p  \left(\frac{p}{ r}\,p_x +
 \frac{q}{ r}\,q_x\right) q_x dx  \\
&  \qquad -  \varepsilon  \int_{\R}  \varphi(r)\,(q_x)^2 dx.
\end{align*}
We then have
\begin{align*}
{\frac{\partial}{\partial t}} \int_{\R} H(r) dx & = \alpha  \int_{\R} \left( g'(r)\,|u|^2 \right)  \varphi'(r)
 \left(p\,p_x + q\,q_x\right)\frac{p^2-q^2}{r^2}\, dx \\
 &+  \alpha  \int_{\R} \left( g'(r)\,|u|^2 \right)  \varphi(r)\,\frac{pp_x-qq_x}{r}dx  \\
&-  \varepsilon  \int_{\R} \varphi'(r)\, r\, (r_x)^2 dx - 
 \varepsilon  \int_{\R} \varphi(r) \left( |p_x|^2 + |q_x|^2\right) dx. 
\end{align*}
Hence, by \eqref{e4.5'},
$$  
{\frac{\partial}{\partial t}} \int_{\R} H(r) dx + 
a\, \varepsilon  \int_{\R} \left( |p_x|^2 + |q_x|^2\right) dx \leq c \,\alpha
 \int_{\R} |u|^2 \left( |p_x| + |q_x|\right) dx, 
 $$
with $c$ independent of $\varepsilon, T$ and $\alpha$.

\end{proof}


 Now, assume $\alpha=O(\varepsilon^{1/2})>0$ and let $\b>0$ be such that 
\begin{equation}\label{e4.9}
 \frac{\a}{\ve^{1/2}}\le \b.  
\end{equation}

{} From \eqref{e4.8} we obtain the following.

\begin{corollary}\label{C:4.4}  Under the hypothesis of the Lemma~\ref{L:4.2}, we have
\begin{equation} 
\int_{\R} r^2 dx +\varepsilon  \int_0^t \int_{\R} 
 (|p_x|^2 + |q_x|^2) \, dx \, d\tau \leq c \, (1 + \beta^2 \int_0^t  \|u_x\|_{2}  d\tau) \label{e4.10}
 \end{equation}
with $c$ independent of $\varepsilon$ and $T$, where $\beta$ is defined by \eqref{e4.9},
\end{corollary}

\begin{proof} Since $\varphi(r)\geq a > 0$, we easily deduce from \eqref{e4.8}, \eqref{e4.9}, 
$$ 
 \int_{\R} r^2 dx +\varepsilon  \int_0^t \int_{\R} 
\,(|p_x|^2 + |q_x|^2) \, dx \, d\tau \leq c \, (1 + \beta^2\int_0^t \int_{\R} \|u\|_4^4 dx \, d\tau). 
 $$
Now we use that $\|u\|_{4} \leq c \, \|u_x\|_{2}^{1/4} \|u\|_{2}^{3/4} $. 

\end{proof}



We next establish  the following analogue of Lemma~\ref{L:2.2}.

\begin{lemma}\label{L:4.4} Let $(u, p,q)$ with $(u,p-p_*,q-q_*) \in C ([0, T) ; H^1({\R}) )^3$, $T > 0$, be a solution of the  Cauchy problem \eqref{e4.1}-\eqref{e4.4}.  Then, there exists $\ve_0>0$ such that if $0<\ve<\ve_0$, we have
\begin{equation}
\| u_x(t) \|_{2}  \leq h(t),   \label{e4.12} 
\end{equation}
where $h\in C^\infty ([0, +\infty))$ is a positive function independent of $T$ and $\varepsilon$.
\end{lemma}

\begin{proof} Define $\theta ( t):= \bigl(1+\max_{\tau\in[0, t]} \|u_x(\tau)\|_{2}^2\bigr)^{1/2}$. {}From \eqref{e4.7} and \eqref{e4.10}, we have
\begin{align*}
\| u_x(t)\|_{2}^2 & \leq c + c\,\varepsilon \alpha \int_0^t \int_{\R} |u|\,|u_x|\, (|p_x| + |q_x|)
\,dx\,d\tau
 +  c\,\varepsilon \alpha \int_0^t \int_{\R} |u|^2\, (|p_x|^2 + |q_x|^2)
\,dx\,d\tau \\
& \leq  c + c\,\varepsilon \alpha  \int_0^t  \|u_x\|_{2}^{3/2} \||p_x| + |q_x| \|_{2}
\,d\tau  +  c\,\varepsilon \alpha \int_0^t  \|u_x\|_{2}\, \||p_x| + |q_x| \|_{2}^2 \,d\tau.
\end{align*}
It follows then that 
\begin{align}
\theta^2(t) & \leq c\,+ c\,\varepsilon\beta  \theta (t)^{3/2} t^{1/2} 
\left( 1 +  \beta   t^{1/2} \theta^{1/2} (t) \right) 
 + c\,\varepsilon^{1/2}\beta \theta (t)(  1 +  \beta^2 \, t\, \theta (t))\notag\\
 &\le c+ c\ve t^{1/2}(1+t^{1/2})\theta(t)^2+ c\ve^{1/2}(1+t)\theta(t)^2. \label{e4.te}
\end{align}
Let $\ve_0>0$ be such that
$$
2c(\ve_0+\ve_0^{1/2})<\frac12.
$$
Then, for $0\le t<1$, from \eqref{e4.te} we deduce 
$$
\theta(t)^2\le 2c
$$
By induction, as in the proof of Lemma~\ref{L:2.2}, we prove that for $n\le t < n+1$, we have
$$
\theta(t)^2\le 2^{n+1}c.
$$
Hence, we get
$$
\theta(t)^2\le 2^{t+1}c,\qquad \text{for $t\ge0$},
$$
which concludes the proof.

\end{proof}


 We finally establish the following result on the global well posedness and convergence of the vanishing viscosity and interaction solutions of
 \eqref{e4.1}-\eqref{e4.4}.

\begin{theorem}\label{T:4.6} There exists $\ve_0>0$ such that for $0<\ve<\ve_0$ the problem \eqref{e4.1}-\eqref{e4.4}
admits a unique global solution $(u^\ve,p^\ve,q^\ve)$ with $(u^\ve,p^\ve-p_*,q^\ve-q_*) \in  C([0, \infty) ; H^1({\R}))^3$. Moreover, as $\ve\to0$ and $\a\to0$ with $\a=O(\ve^{1/2})$, we may extract a subsequence of solutions $(u^\ve,p^\ve,q^\ve)$ converging in $L_\loc^1(\R\X[0,\infty))$ to some $(u,p,q)\in C([0,\infty);H^1(\R))\X L^\infty(\R\X[0,\infty))^2$ which is a weak (entropy) solution of
\begin{align}
& iu_t+u_{xx}=u|u|^2, \label{e4.1'}\\
& p_t+(\vph(r)p)_x=0, \qquad r=\sqrt{p^2+q^2}, \label{e4.2'}\\
& q_t-(\vph(r)q)_x=0.\label{e4.3'}\\
&(u(x,0),p(x,0), q(x,0))=(u_0(x),p_0(x),q_0(x))\label{e4.4'}.
\end{align}

\end{theorem}

\begin{proof}  1. The existence and uniqueness of a global solution of \eqref{e4.1}-\eqref{e4.4}, $(u^\ve,p^\ve,q^\ve)$, with $(u^\ve,p^\ve-p_*,q^\ve-q_*) \in  C([0, \infty) ; H^1({\R}))^3$ follows from the estimates provided by  Lemmas~\ref{L:4.1}--\ref{L:4.4}, which allow the extension of the unique local in time solution, as in the earlier sections. 

2. The compactness of $\{u^\ve\}$ follows as before from Aubin's lemma, in view of the referred estimates. The fact that the limits of converging subsequences must belong to $C([0,\infty);H^1(\R))$ is a consequence of the uniform boundedness of $u^\ve$  in $L_\loc^\infty([0,\infty);H^1(\R))$ and Duhamel's formula.

3. As to the compactness of $(p^\ve,q^\ve)$ in $L_\loc^1(\R\X(0,\infty))$, since $\a=O(\sqrt{\ve})$ the estimates given by  Lemmas~\ref{L:4.1}--\ref{L:4.4} give that
the term $\a(g'(r)|u|^2)_x$ is  uniformly bounded in $L_\loc^1(\R\X(0,T))$. Thus, using those estimates, we deduce that for any entropy-entropy flux pair for \eqref{e4.2'},\eqref{e4.3'}, $(E(p,q),F(p,q))\in C^2((0,\infty)\X(0,\infty))^2$, we have 
\begin{equation}\label{e4.Se}
E(p^\ve,q^\ve)_t+F(p^\ve,q^\ve)_x\in \{\,\text{compact of }\,W_\loc^{-1,2}(\R\X(0,\infty))\,\},
\end{equation}
which allows us  to apply the compactness result in \cite{Se2}.


\end{proof} 





\section{SW-LW interactions with general symmetric systems} \label{S:6}

In this section we consider a general symmetric system of conservation laws, that is,
the vector flux-function is the gradient of a real-valued $C^3$ function
$\varphi (v)$, $v = (v_1, \ldots, v_n) \in {\R}^n$. We propose the following
model for SW-LW interactions
\begin{align}
&  i u_t + u_{xx} = |u|^2 u +\alpha\, g(v)\, u    \label{e5.1}   \\
& v_t + \left(\nabla_v \varphi (v)\right)_x =   \alpha\, \left(\nabla_v g(v) |u|^2 \right)_x +
 \varepsilon \, v_{xx},\quad \alpha > 0,\, \varepsilon > 0.   \label{e5.2}
\end{align}
For simplicity we will assume that the corresponding nonlinear symmetric system of conservation laws with viscosity
\begin{equation}
 v_t + \left(\nabla_v \varphi (v)\right)_x =\ve v_{xx},\label{e5.2B}
\end{equation}
admits a bounded invariant domain $D$, which is the case of the systems considered in \cite{CK1,CK2, FS1}, for instance.  

Concerning $g$, we assume that it is a real-valued $C^2$ function   such that $g'$ has  support contained in the interior of the invariant domain $D$. 

We assume $\nabla_v \varphi (0) = 0$ and that $\nabla_v^2 \varphi (0)$ (Hessian
matrix of $\varphi$ at $v=0$) is equal to the $n\times n$ zero matrix. 
We consider the Cauchy problem for \eqref{e5.1},\eqref{e5.2} with initial data given by
\begin{equation}\label{e5.3}
(u(x,0),v(x,0))=(u_0(x0,v_0(x)),
\end{equation}
with
\begin{equation} 
(u_0,v_0-v_*)\in H^1 ({\R})^{n+1},\label{e5.3B} 
\end{equation}
where $v_*$ is a point in the interior of the invariant domain $D$ and $v_0(x)\in D$ for all $x\in\R$. 

Exactly as in Lemma~\ref{L:4.1} we obtain the following estimate.

\begin{lemma}\label{L:5.1} Let $(u,v)$ be a local in time solution of \eqref{e5.1}-\eqref{e5.3} with $(u,v-v_*)\in C([0,T);H^1(\R))$, for some $T>0$. Then
\begin{equation} 
| v(t) | _{L^\infty ({\R})} \leq c   \label{e5.4} 
\end{equation}
where $c$ is independent of $\ve,t, T$.
\end{lemma}

Using \eqref{e5.4} and the symmetry of \eqref{e5.3} we obtain, with essentially the same proofs, the analogues of lemmas are similar to Lemmas~\ref{L:2.1} and \ref{L:2.2}, which we only  state below for the sake of reference.   

\begin{lemma}\label{L:5.2}  Let $(u, v) \in C ([0, T) ; H^1({\R}) )^{n+1}$, $T > 0$, be a solution of the Cauchy problem \eqref{e5.1},\eqref{e5.2},\eqref{e5.3}. We have:
\begin{align}
  & {\frac{\partial}{\partial t}} \int_{\R} |u|^2 dx = 0.      \label{e5.5}   \\
   &  \notag \\
 & {\frac{\partial}{\partial t}}\, \left[ {\frac12} \int_{\R} |u_x|^2 dx +  
 {\frac14} \int_{\R} |u|^4 dx +   {\frac{\alpha} 2} \int_{\R} g(v)\, |u|^2 dx  
   - {\frac12 }   \int_{\R} \varphi (v) dx \right] \label{e5.6}   \\
     & \qquad + \frac{\varepsilon\alpha}{ 2}  \int_{\R} ( |u |^2)_x \nabla_v g(v)\cdot v_x  dx +
 \frac{\varepsilon\alpha}{ 2}  \int_{\R}  |u |^2 (v_x^\Tr\, \nabla_v^2 g(v)\,v_x)\, dx \notag  \\
 & \qquad - {\frac\varepsilon 2 }  \int_{\R}v_x^\Tr\, \nabla_v^2  \varphi (v)\,v_x\, dx  = 0.\notag\\
   &   \notag \\
 &  {\frac{\partial}{\partial t}} \left[{\frac12 } \int_{\R} |v|^2 dx +
 {\rm Im} \int_{\R} u \bar u_x  dx \right] +  \varepsilon   \int_{\R} |v_x|^2 dx = 0.   \label{e5.7}
\end{align}
\end{lemma}


\begin{lemma}\label{L:5.3}  Assume \eqref{e5.4} and suppose that 
$\a=O(\varepsilon^{1/2})$. Under the hypothesis of the Lemma~\ref{L:5.1} there exists $ \varepsilon_0 > 0$,
independent of $T$, such that for $\varepsilon \leq \varepsilon_0$
\begin{equation} 
| u_x |_{L^2({\R})} (t) \leq h (t), \label{e5.9} 
\end{equation}
where $h\in C^\infty ([0, +\infty)) $ is a positive function independent of $T$ and $\varepsilon$.
\end{lemma}

As in the earlier section, using the estimates given by Lemmas~\ref{L:5.1}--\ref{L:5.3} we arrive at the following result. 

\begin{theorem}\label{T:5.1}  Let $\a=O(\ve^{1/2})$. There exists $\ve_0>0$ such that
for  $\varepsilon \leq  \varepsilon_0$ there is a unique global solution
 $ (u^\varepsilon, v^\varepsilon) \in \left(C ([0, +\infty[ ; H^1({\R})) \right)^{n+1}$
of the Cauchy problem \eqref{e5.1},\eqref{e5.2},\eqref{e5.3}.  Moreover, there exists $ u\in L_{\loc}^\infty
([0, +\infty[ ; H^1({\R}))$ and a subsequence of $\{u^\varepsilon\}_\varepsilon$ converging in $L_\loc^1(\R\X[0,\infty))$ to $u$. Further, if uniformly bounded solutions of \eqref{e5.3B} as $\ve\to0$  form a precompact sequence in $L_\loc^1(\R\X(0,\infty))$ (as in \cite{CK1,CK2,FS1}, for instance),then $\{v^\ve\}$ is precompact in $L_\loc^1(\R\X(0,\infty))$ and any limit of a convergent subsequence of $v^\ve$ is an entropy solution of the symmetric nonlinear system of conservation laws obtained from \eqref{e5.2} when $\a=\ve=0$.   
\end{theorem}


\begin{remark} We observe in conclusion that some symmetric systems with unbounded invariant domains, such as the one in \cite{FS2},  admit  a completely similar result. 
\end{remark}

\section*{Acknowledgements} The research of J.-P.~Dias and M.~Figueira was partially supported by FCT under program POCI (Portugal/ FEDER-EU). H.~Frid gratefully acknowledges the support of CNPq, grant 306137/2006-2 ,
and FAPERJ, grant E-26/152.192-2002.   
   

 
   






\end{document}